

# CASPER WEB USER MANUAL

Learn how to submit data via the Centralised Submission Platform

© 2022 | European Central Bank | CASPER@ecb.europa.eu

### Contents

IN	ITROD		3
1	GEN	IERAL DESIGN PATTERNS AND APPLICATION BEHAVIOUR	4
	1.1	CASPER landing page	4
	1.2	CASPER navigation	7
	1.3	Move and resize pop-ups	10
	1.4	List views, filtering and sorting	10
	1.5	Export functionality	19
2	нои	V TO ACCESS CASPER	20
	2.1	Login (authentication process)	20
	2.2	User roles and permissions	21
3	USE	R JOURNEYS	23
	3.1	Journey I: web user administrator	23
	3.2	Journey II: web data submitter	29
A	PPEND	אות	72
	Figure	s	72
	Tables	; 74	
	Abbre	viations and terminology	75

### **INTRODUCTION**

The Centralised Submission Platform (CASPER) allows external organisations and partners to submit structured data securely to the European Central Bank (ECB). The data are automatically validated, and the results can be discussed with the relevant ECB teams within the platform.

The browsers supported by CASPER are Google Chrome and Microsoft Edge.

If you have an account, you can sign in here:

### https://casper.ecb.europa.eu/casper-internet

This user manual provides information on how to best use CASPER to submit data to the ECB, monitor fulfilment of reporting obligations and administer users of the system.

The document starts by introducing the general design patterns and application behaviour applicable to the main parts of the system (Chapter 1), followed by the different user roles and corresponding permissions (Chapter 2). Subsequently, it explains how each user group undertakes its user journey (Chapter 3).

## 1 GENERAL DESIGN PATTERNS AND APPLICATION BEHAVIOUR

In this chapter, you will find a detailed description of CASPER and its behaviour, navigation and general design patterns.

### 1.1 CASPER landing page

Once you have successfully logged into CASPER, you will see the CASPER landing page (Figure 1).

You can also access the page from every CASPER screen by clicking on the logo at the top.

On the landing page you will see the announcements (Box 1 in Figure 1) and the privacy statement (Box 2). The privacy statement is positioned at the bottom of the page, but you can access it by clicking on the "Privacy Statement" link (Box 3) in the upper right-hand corner of the screen.



Figure 1: CASPER landing page

### 1.1.1 Announcements

On the landing page you will see the Announcements section.

Here you will find communications from the system administrators or collection owners (for detailed descriptions of these, please refer to the Abbreviations and terminology section in this manual). The announcements you see on the landing page will change over time as new ones are posted and older ones expire or are deleted.

You will see a preview of all the announcements that are relevant to you.



Figure 2: Announcement preview sample

For each announcement, the following elements are displayed (as shown in Figure 2).

- 1. The title of the announcement: if the length of the title exceeds the width of the announcement, hover your mouse cursor over the title to view the full text as a tooltip.
- 2. A flag, which can be either "General" or a data collection code. General announcements are visible to all users; announcements specific to data collections are visible only to users authorised to access those collections.
- 3. The description of the announcement; only the first 150 characters are shown, and the formatting of the announcement is not visible.
- 4. A "Click here" button enabling you to see the full details of the announcement.

General announcements are posted by the CASPER administrators and provide, for example, information about new releases, links to documentation and messages about scheduled maintenance dates.

In announcements specific to data collections you will find relevant information about the collections that you are authorised to access. If you click on the "Click here" button in one of the announcements, a pop-up window will appear, as shown in Figure 3. The window will display the full text of the announcement.

Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  Trite:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  TRITE:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER  TRITE:PANNA COTTA RELEASE WILL GO LIVE IN CASPER PRODUCTION IN OCTOBER
DearCASPER users, to inform you about ongoing activities in CASPER, the team is currently performing regression tests in order to prepare the planned move of Panna Cotta into production. Exact dates will be communicated to you in due time. Best regards, Your CASPER team
Close

### Figure 3: Announcement pop-up window

Table 1 shows the fields displayed in the pop-up window.

Field	Description
Title	The title of the announcement.
Data collection code	Only displayed when the announcement is for a specific data collection, otherwise the value shown is "General".
Description	The full text of the announcement, as formatted by the user who posted it.

Table 1: Announcement pop-up fields

You can only see announcements specific to data collections for collections that you are authorised to access. General announcements are visible to all users.

### 1.1.2 Manage email subscription

CASPER generates email notifications to inform users about established events in the collection lifecycle. By default, users do not receive email notifications. For email notifications to be displayed, the function must be enabled in the notifications tab by the internal collection designer (more information about this role can be found in the Abbreviations and terminology section). The users assigned to the relevant data collection can then set up a subscription in the manage email subscription screen. To view this screen, log into CASPER and click on the mail symbol in the upper right-hand corner of the screen (Figure 4).

EUROPEAN CENTRAL BANK   EUROSYSTEM Centralised Submission Platform (CASPER)	
Submissions	
Announcements	

Figure 4: CASPER landing page

Box 1 in Figure 4 shows the email subscription icon. Click on this icon to open the email subscription screen.



Figure 5: Receive notifications as emails

1. Click on the drop-down list and select "Yes".

You can also set up an email subscription in the collections screen by right clicking on the data collection assigned to you and selecting "Manage email subscription".

Figure 6 shows the manage email subscription screen. The list on the screen displays all available email notifications for the data collections assigned to you.

			EUROPEAN CENTRA	L BANK   EUROS	YSTEM		Enveronment Acce
	Manage Em	ail Subscri	ption				oles .
	Receive notifications	as emails	Your email address				
	Yes		- Management of the				
	Announcements		Immediately	🕖 Daily	U Weekly	Never	
	Data Collection		Data Collection Name				
4	Research		Record Scrapping				
	Receiving potific for the individua	ations as emails a I event type.	iso cepends on the collect	ion owner to activat	e the creation of noti	fication emails	
	✓ Metadat	a Import Notific	ation				

Figure 6: Manage email subscription screen

- 1. The "Receive notifications as emails" list indicates whether the email subscription is active. Click on the collapsible menu and select "Yes" or "No" to activate or deactivate it.
- 2. The "Your email address" tab displays the recipient's email address.
- 3. Set the announcements radio button to "Immediately", "Daily", "Weekly" or "Never" to adjust how often email notifications are received when announcements are posted on the CASPER landing page.
- 4. Select the data collection from the drop-down list to enable email notifications for specific events.

Figure 7 shows how to enable email notifications for specific events for a data collection.

	Manage Email Subscription					
Data Collection	Receive notifications as emails	Your email address				
	Ves 4	Research to the second				
none strents 1	Announcements	) immediately	O Daily	Weekly	Never	Reset view
COD4 *	Data Collection	Oata Collection Name				# ACTIONS
lotter.	· financial second ·	Automation and the				
	<ul> <li>Metadata Import Notification</li> </ul>					
2	Submission Notification					
2		immediately	🖯 bely	U Workly	Never	
2	Submission Notification	3 investury	O Dely	U Weekly	<ul> <li>New</li> <li>New</li> <li>New</li> </ul>	
2	Submission Notification Submission					
2	Submassion Notification Submassion Attachment Upsiad Succeeded	3 O mmediately O mmediately	O Dely	O Weekly	Never	

Figure 7: Email subscription configuration

- 1. Select the data collection from the drop-down list to enable email notifications for specific events.
- 2. To manage email subscription for specific events, first open the collapsible panel, then set the frequency by clicking on the required radio button. You can set the email subscription for multiple notifications.
- 3. Set the notification frequency radio button to "Immediately", "Daily", "Weekly" or "Never" to adjust the frequency with which email notifications are received. (The default value is "Never".)
- 4. Set a bulk email subscription frequency for a specific group of notifications by selecting the frequency header at the top of the screen.

### **1.2 CASPER navigation**

In this section, you will be introduced to CASPER's main navigation function and learn how it works. Figure 8 shows the main navigation screen, which has three parts: the primary navigation pane, the secondary navigation pane and the vertical navigation pane.

<u> </u>			Subr	nissio	ns					User	& Role	s		
Information	=		ion Lis				Refresh	Download Files		Zip Upload	Frm	ort CSV	Rese	tView
Submission List File Vault		. =	CYCLE	=	REPORTING	1*=	MODULE 2 *=			REFERENCE DATE	=	OBLIGATION STATUS	=	RECEPTION
Validation Results		۲ 🗌		۲		۲	τ		۲	O DD / MN/ YYYY	۲		۲	
Discussions		~ 5	RD2		ENTITY1		CSV	Entity1		19/11/2021		0		19/11/202
Reports		~ 8	RC1		ENTITY1		EXCELWEB	Entity1		18/10/2021		A		19/10/202

#### Figure 8: Main navigation screen

- 1. **Primary navigation pane:** This is the top section of the main navigation screen. It has a light blue background and covers the whole width of the screen.
- 2. **Secondary navigation pane**: This is the lower section of the navigation screen. It has a dark blue background and extends over the whole width of the screen.
- 3. **Vertical navigation pane:** This appears on the left-hand side of the screen when you click on a secondary navigation item. The number of menu items differs according to the secondary navigation item selected.

Figure 9 shows the five items highlighted in the primary navigation pane. Table 2 describes these items in more detail.



### Figure 9: Primary navigation pane

Item	Description
Top logo ("EUROPEAN CENTRAL BANK   EUROSYSTEM") (1)	This is clickable and redirects the user to the CASPER landing page (see Section 1.1).
Notifications button (2)	This is displayed as a mail icon with a number indicating the number of unread notifications in the inbox. Once the number of notifications exceeds 99, "99+" is displayed. By clicking on this icon, you will be redirected to the notifications screen.
Email subscription button (3)	This is displayed as a mail icon. By clicking on this icon, you can enable/disable notifications sent by email.
Logout (4)	By clicking on the "Logout" button, the user is redirected to the login page.
Language drop-down menu (5)	Users can switch between the available languages by clicking on the drop-down menu arrow. Currently, the entire CASPER platform is only available in English.

#### Table 2: Primary navigation items

Figure 10 shows the secondary navigation items, with each item described in Table 3. For a more detailed description, please refer to the Abbreviations and terminology section.



### Figure 10: Secondary navigation pane

To enable access to CASPER, a role is assigned to a user with a predefined set of permissions. Depending on the role, the user will see one or more items in the secondary navigation pane, i.e. some users may see both items, whereas others may see only one.

Item	Description
Submissions (1)	Visible to web roles that use the platform to monitor reporting obligations and to external roles that submit files, monitor processing and discuss failed validation results. For further information, see Section 3.2.
User & Roles (2)	Visible to every user administrator. User administrators can view and manage the assignment of other users to data collections using the User & Roles link. For further information, see Section 2.

### Table 3: Secondary navigation items

If one or more of the navigation items described in the table above are missing, this indicates that your role does not have permission to access the information.

Each secondary navigation item has its own set of vertical navigation menus located on the left-hand side of the screen.

Figure 11 and Figure 12 show the vertical navigation menu for the "Submissions" item of the secondary navigation pane. The highlighted steps can be followed to navigate to any vertical navigation pane.



### Figure 11: Access vertical navigation





- 1. Click on the "Submissions" tab in the secondary navigation pane to access the submissions overview screen.
- 2. Select the desired data collection from the list by clicking on the Actions menu and selecting "View submission list". The selected collection will now be displayed in the upper left-hand corner of the submission list screen.
- 3. Once you have selected the data collection, the vertical navigation menu will appear on the left-hand side of the screen. The menu item changes to blue when you select it.

If there is a small arrow beside a menu item or any other element, this means that itis collapsible. Click on the small arrow to expand the list and click on it again to collapse the list (see Figure 13).

		EU	ROPEAN CENTRAL	BANK   EUROSYSTEM		Environr
		c	Centralised Submission	n Platform (CASPER)		On Logout EN
<u>,</u>				User & Roles		
	User - Data Co	ellection Assignment	List			
Users -	Active Search					
Users • User Assignment User Entity Access	Active Search	USER NAME	USER ROLE	DATA COLLECT	ONS	

Figure 13: Collapsible menu

### 1.3 Move and resize pop-ups

Some of the pop-ups in the application can be moved or resized, as illustrated in Figure 14.

lect	Marina		CORCEOID			X	
	User ID		User Name		Country	1	
	User Role						Searc
64	DATA COLLECTIC	ONS					
SER M	=	CODE		=	NAME		SIGN
otron otron		at 611_64101 ***6611		T		Save Cancel	2

Figure 14: Movable and resizable pop-ups

- 1. **Moving a pop-up:** To move a pop-up, place the mouse cursor in the top section of the screen (Box 1). If the pop-up is movable the mouse cursor will turn into a 4-arrow symbol. By left clicking you will be able to drag the pop-up around the screen.
- 2. **Resizing a pop-up:** To resize a pop-up, place the mouse cursor in the lower left-hand corner of the screen (Box 2). If the pop-up is resizable the mouse cursor will turn into a 2-arrow symbol. By left clicking you will be able to resize the pop-up, keeping the aspect ratio.

### 1.4 List views, filtering and sorting

CASPER displays content in the form of lists. This is one of the most frequently used features in the application. This section describes the design, functions and behaviours of lists. Figure 15 shows an example of a list. An illustration of

its five main parts and a detailed description of each one is given below. Further details are provided in the following sub-sections.



#### Figure 15: List components

- 1. **Table header:** The header for each column in the list can be found here. Type a word in the textbox that is displayed under each header to filter items.
- 2. **Sorting:** The list has a sorting functionality which allows you to arrange the elements in the list in ascending or descending order by column. Simply click on the column header label once and this will sort the rows. See Section 1.4.3 for a detailed explanation of the sorting options available.
- 3. **Filter button:** Type a word in the textbox or click on the small "Filter" button on the right-hand side of the screen to filter rows and information. Section 1.4.2 explains the filter options in greater detail.
- 4. **Actions:** If you click on the "Actions" button a pop-up window will appear with a list of action items you can perform for each entry in the list. See Section 1.4.6 for further details.
- 5. **Customisation panel**: Using this panel you can adjust the column size, add or remove toggle columns, etc. To close the customisation panel, click on the customisation panel button again or click anywhere on the screen. See Section 1.4.1 for further details.

### 1.4.1 Customising list columns

You can customise the appearance of the list using the customisation panel. There are many customising functionalities available, such as adding toggle columns, adjusting the column width or pinning selected columns on one side. This section describes all the available features of the customisation panel.

The customisation panel button is located in the table header and in every column, as shown in Figure 16.

Number of Entries for	und: 57			
CODE		123 = T =		
NTST2	test	Pin Column Autosize This Column Autosize All Columns	Pin Left Pin Right	
OPTBL1	test	Reset Columns Tool Panel	✓ No Pin	
OTTST1	test	test		
оттятз	test	test		

#### Figure 16: Customisation panel

When you click on the customisation panel button, a pop-up menu with options will appear. On the upper left-hand side of the pop-up menu, you will see three submenu buttons showing different options, as described below.

1. General options: change the appearance of the columns.

- 2. Filter pane: filter the information in the columns.
- 3. Toggle columns: add or remove columns in the list.

You can also customise the order in which the columns are displayed by using the drag-and-drop functionality: click on the desired column, then hold down and move the mouse to the desired position in the list. Please note that all changes will be lost if you leave this screen. The default settings will be shown when you return.

### 1.4.1.1 General options

The general options in the first submenu in the customisation panel are used to customise the columns. Figure 17 shows the option items in the first submenu. Table 4 contains a description of each item.

	found: 57					
CODE	≡	NAME 🗸				=
NTST2	T	test	Pin Column Autosize This Column Autosize All Columns	Pin	ı Left ı Right	T
OPTBL1		test	Reset Columns Tool Panel	✓ No	Pin	
OTTST1		test	test			

#### Figure 17: General options submenu

Item	Description
Pin column	Pins a column of your choice to either the left or right-hand side of the list. If you select "No pin", the column is displayed in its default position in the list.
Autosize this column	Automatically adjusts the width of a specific column, based on the amount of text it contains.

Item	Description
Autosize all columns	Automatically adjusts the width of all the columns.
Reset columns	Undoes all changes and resets to the default view for the list.
Tool panel	Pins the tool panel window on the right- hand side of the list. By default, it displays all the toggle columns. In the "Tool panel" window, you can add columns to the list or remove them by ticking the checkbox for each column.

Table 4: General options items

### 1.4.1.2 Filter pane

The second submenu in the customisation panel is the filter pane. This provides features to filter rows based on the filter options available for the list. You can access the filter pane from this submenu or from the filter button, as described in Section 1.4.2. Either way, the functionality remains the same. See Section 1.4.2 for further details.

### 1.4.1.3 Toggle columns

For most screens in CASPER, there are columns which are not visible in the list by default. The "Toggle column" button enables you to view the columns that are available and add them to the list as required. The list of columns available in the toggle column feature varies according to the screen.

The following steps explain how to add columns to the list, as shown in Figure 18, 19 and 20.

- 1. Click on any of the highlighted customisation panel buttons, as shown in Figure 18. A pop-up menu will appear which shows three submenu buttons; the one furthest to the left is selected by default.
- 2. Click on the "Toggle columns" button, as shown in Figure 19. This displays a list of all available columns. The columns displayed in the list by default are checked with a blue tick; the remainder are unchecked.
- 3. To add a column to the list, select it by ticking the corresponding checkbox, as shown in Figure 20. Once you have selected a column, it is added to the list instantly. You can add multiple columns at the same time.
- 4. To remove a column from the list, untick the corresponding checkbox, as shown in Figure 20. Once you untick a column, it is removed from the list instantly. You can remove multiple columns at the same time.
- 5. Once you have added or removed the columns from the list, click anywhere on the screen to leave the toggle column menu.



Figure 18: Adding toggle columns 1/3

Numb	er of Ent	ries found: 404											
	# =	CYCLE	=	REPORTING	1 ≛≣	MODULE 2*	=	ENTITY NAME	=	REFERENCE DATE	-		R
	<b>T</b>		۲		т		۲		۲	OD/MN/YYYY		Filter	2
	~	-								18/12/2020		CYCLE REPORTING CODE	1
	~									26/12/2020		MODULE ENTITY NAME	41
	~									27/12/2020		REFERENCE DATE OBLIGATION STATUS	
	~									25/12/2020		RECEPTION DATE REMITTANCE DATE	
										18/12/2020		ATTACHMENT DELAY IN DAYS	
	~									27/12/2020	P	FAILED ATTEMPTS SEQUENTIAL NUMBER	

#### Figure 19: Adding toggle columns 2/3

Numb	er of Ent	ries found: 404												les	Zip Uple	oad	
	. =	ENTITY NAME	=	REFERENCE DATE	-	OBLIGATION STATUS		RECEPTION 3 -= DATE	REMITTAN	E T 1	4		DELAY IN DAYS	=	FAILED ATTEMPTS	-	астіо
	۲		۲	OD/MN/YYYY	۲		r		O DD / MI	L] THEET	4	۲		*		*	3
									31/12/202	_ KEI OKIIIKO CO	DE	1			1		1
									31/12/20	MODULE ENTITY NAME		1			0		1
								-	31/12/202	<ul> <li>REFERENCE DATION ST</li> <li>OBLIGATION ST</li> </ul>	ATUS	1			0		1
									31/12/202	RECEPTION DA	TE ATE	3			1		1
										DELAY IN DAYS			161		0		1
									31/12/202	FAILED ATTEMP	TS				0		

Figure 20: Adding toggle columns 3/3

While you are logged in (or if you have logged out but the browser is still open), CASPER will retain the changes made to each screen using the tool panel. You can return to the default settings by clicking on "Reset view".

### 1.4.2 Row filtering

You can filter column content using the filter boxes, as displayed in Figure 21. You will find an explanation of the filter boxes and the other highlighted sections below.

hold B	Sub	mis	sic	on List							
Submission List	Num	ber of	Entri	es found: 404							1
File Vault Validation Results		#	=	CYCLE	=	REPORTING	1 ≛≣	MODULE 2 *	ENTITY NAME	=	REFERENCE DATE
Discussions			۲		۲	2	۲	T		۲	OD/MV/YYYY
Reports		$\sim$	4								18/12/2020
		$\sim$	11								26/12/2020
		$\sim$	8	-							27/12/2020

### Figure 21: Table headers

In most columns, you will see an extra field below the header label (1). This may be a textbox (2) or a date field (3). The purpose of the additional fields is to give you more flexibility when filtering row entries in the list. You can filter results by entering text in the textbox or choosing a specific date in the date field.

On the top left-hand side of the list, you will see the summary text. This is a dynamic field showing the total number of row entries in the list.

The textbox shows text in lower case. If you enter text in upper case, it will be automatically converted to lower case. The date field only accepts numeric entries.

The filter button provides more options for filtering the rows in the list. For every column in the list (except "Actions"), you will see a small filter button, as highlighted in Figure 22. When you click on this button, a small pop-up window will appear, with two fields: a drop-down list and a textbox.



Figure 22: Filter button pop-up window

The items in the drop-down list depend on the column data type, i.e. they are different for columns with text content and columns with date content.

Numb	er of I	Entrie	es found: 404												Downlos	nd Files	Zip Uplo
	•	=	CYCLE	=	REPORTING	1*=	MODULE 2 * =	ENTIT	TY NAME	=	REFERENCE DATE	OBLIC	SATION	=	RECEPTION	3 ⊷≣	REMITTANCE
		۲		٣		T				۲	т			۲			D / MN / YY
	~	4	RC1		ENTITY1		ntains	v	Y1		Equals Equals	v	A		Select All)	Delaved	12/2020
	$\sim$	11	RC1		ENTITY1	No	n <mark>tains</mark> t contains		Y1		Greater than Less than		A		Not Received	Not Due	
	~	8	RC1		ENTITY1	No	uals t equal		771		Not equal In range		A		Received - Err	or	12/2020
	~	3	RC1		ENTITY1		rts with ds with		7Y1		25/12/2020		A	Y	A Received - Wa Revalidated -	Error	12/2020
		0											Ø		Revalidated - I	Fatal	12/2020
-	~	8											A		A Revalidated -	Warning	12/2020

#### Figure 23: Filter pop-up types

Figure 23 shows the different types of filter pop-ups for different column types in the data collections screen. The filter pop-ups for other screens display different statuses.

There are different filter pop-ups for columns which display the status of a row entry. These columns have predefined values that you can filter for using the filter pop-ups. The possible status types differ, depending on the screen you are viewing.

For columns with text content, when you select an item from the drop-down list you can use the textbox underneath to filter information. For example, if you select "Contains" from the drop-down list and write "a" in the textbox underneath, the list will filter for the rows which **contain** an "a" in them, as shown in Figure

### 24.



Figure 24: Example of a text filter

While you are logged in (or if you have logged out but the browser is still open), CASPER will retain the changes made to each screen using the tool panel. You can return to the default settings by clicking on "Reset view".

### 1.4.3 Sorting

Lists in CASPER have built-in sorting functionalities. Each screen has a specific pre-set default sorting order. Sorting is available for every column in the list.

You can sort a column by clicking on its header label. When you do this, a small blue arrow (pointing upwards or downwards) will appear next to the label. The

arrow indicates the sorting order, either ascending or descending (see Figure 25 and Figure 26).



### Figure 25: Ascending sorting order

Subi	mis	sic	on List									
	_											
Numb	er of	Entri	es found: 404									
	#	=	CYCLE	=	REPORTING CODE	=	MODULE	=	ENTITY NAME	=	REFERENCE DATE	=
		۲		T		T		Ŧ		۲	OD/MN/YYYY	1
		0	RC1		ENTITY99				ENTITY99		18/12/2020	
	$\sim$	6	RC1		ENTITY99				ENTITY99		27/12/2020	
		0	RC1		ENTITY99				ENTITY99		26/12/2020	
		0	RC1		ENTITY99				ENTITY99		25/12/2020	
		0	RC1		ENTITY98				ENTITY98		26/12/2020	
		0	RC1		ENTITY98				ENTITY98		18/12/2020	

#### Figure 26: Descending sorting order

You can also sort list entries by multiple columns. To do this, click on the first column header. Then press and hold down the Shift key and click on the

remaining column headers you would like to add to your multiple sorting. The numbers to the right of the column name indicate the sorting order.

When you add a column in multiple sorting you need to establish its sorting order (ascending/descending). The sorting order of the first column serves as a reference point for the multiple sorting – if the first column is selected as descending, columns added subsequently will be sorted in descending order too. To change the sorting order of an additional column to ascending, click on the selected column a second time. The sorting order of a column needs to be set before you add another column. It is not possible to change the sorting order of a column in multiple sorting once you have added another column.

Figure 27 shows the multiple sorting of a list by three columns. The list entries are sorted in ascending order by reporting code (1), reference date (2) and reception date (3). The following steps describe how to sort a list by multiple columns in ascending order.

ubr	nis	sic	on List									
Numb	er of I	Entrie	es found: 404									Download File
		=	CYCLE	=	REPORTING 1 *= CODE	MODULE	=	ENTITY NAME	=	REFERENCE 2 *= DATE	OBLIGATION STATUS	RECEPTION 3
		۲		т	τ		۲		۲	OD/MV/YYYY T		τ
	$\sim$	4	RC1		ENTITY1			ENTITY1		18/12/2020	A	23/04/2021 16:00:34
	$\sim$	3	RC1		ENTITY1			ENTITY1		25/12/2020	A	20/12/2020 16:11:34
	$\sim$	11	RC1		ENTITY1			ENTITY1		26/12/2020	A	22/04/2021 12:47:15
	$\sim$	8	RC1		ENTITY1			ENTITY1		27/12/2020	A	12/04/2021 17:24:53
		0	RC1		ENTITY10			ENTITY10		18/12/2020	0	
		1	RC1		ENTITY10			ENTITY10		25/12/2020	A	19/12/2020 13:32:13

Figure 27: Multiple sorting

- 1. Click on the column header of the first variable, "Reporting code".
- 2. Press and hold down Shift and click on the column header of the second variable, "Reference date". Small numbers will appear next to the column headers, indicating the sequence of the order. The small arrows indicate that the order is ascending, as set by default.
- 3. Continue to hold down Shift and click on the column header of the third variable, "Reception date". A small number will appear, indicating that this is the third variable in the multiple sorting, also sorted in ascending order.
- 4. The columns are now sorted by these three variables, with the reporting code in ascending order, the reference date in ascending order and the reception date in ascending order. To undo the sorting, release Shift and click anywhere on the list. The list will return to the default sorting order.

To sort the list in descending order, follow the same steps. First, change the sorting of column one from ascending order by default to descending. To do this, simply click on the column header as described above. Once the arrow appears, add the remaining columns to your multiple sorting order as described above.

Figure 28 shows the list entries sorted by the same columns by reporting code (1), reference date (2) and reception date (3). This time, however, the reference dates are sorted in descending order. The following steps describe how to change the sorting order of a column.

Subr	nis	sic	on List											
Numb	er of l	Entrie	es found: 404										Download File	es
	#	=	CYCLE	=	REPORTING 1 TE	MODULE	=	ENTITY NAME	=	REFERENCE 2 -= DATE	OBLIGATION STATUS	≡	RECEPTION 3 · DATE	•=
		۲		۲	۲		۲		۲	OD/MV/YYYY T		۲		۲
	$\sim$	8	RC1		ENTITY1			ENTITY1		27/12/2020	A		12/04/2021 17:24:53	
	$\sim$	11	RC1		ENTITY1			ENTITY1		26/12/2020	A		22/04/2021 12:47:15	
	$\sim$	3	RC1		ENTITY1			ENTITY1		25/12/2020	A		20/12/2020 16:11:34	
	$\sim$	4	RC1		ENTITY1			ENTITY1		18/12/2020	A		23/04/2021 16:00:34	
	$\sim$	8	RC1		ENTITY10			ENTITY10		27/12/2020	A		12/04/2021 17:27:26	
	$\sim$	8	RC1		ENTITY10			ENTITY10		26/12/2020	A		12/04/2021 17:27:01	

Figure 28: Multiple sorting with column in descending order

1. Click on the column header of the first variable, "Reporting code".

- 2. Press and hold down Shift and click on the column header of the second variable, "Reference date". Small numbers will appear next to the column headers, indicating the sequence of the order. The small arrows indicate that the order is ascending, as set by default. To change the sorting order, click on the column header again. The arrow will now point downwards, indicating that the order for this column is descending.
- 3. Continue to hold down Shift and click on the column header of the third variable, "Reception date". A small number in brackets will appear, indicating that this is the third variable in the multiple sorting, also sorted in ascending order.
- 4. The columns are now sorted by these three variables. To undo the sorting, release Shift and click anywhere on the list. The list will return to the default sorting order.

Rows containing values that start with a number are sorted before those that start with characters.

When you add a new row to the list, the system highlights it in blue and the sorting order remains the same, as shown in Figure 29. When an entry is deleted from a list that has already been sorted, the sorting order remains the same – only the deleted entry is removed.

CODE 🔺	=	NAME	≡	DESCRIPTION	1
	۲		۲		
-	-	Research to 1	1000	UAT Mar 2019	
2TSTPE					
AAA		AAA		AAA	
ABC		ABC		ABC	
В		В		asd	

Figure 29: Example of an added row

While you are logged in (or if you have logged out but the browser is still open), CASPER will retain the changes made to each screen using the tool panel. You can return to the default settings by clicking on "Reset view".

### 1.4.4 Search panel

For some lists, you can also filter for specific entries using the search panel. This allows you to apply multiple filters simultaneously, as displayed in Figure 30.

USER ID	USER NAME		USER ROLE			DATA COLLECT	IONS				2	. )				
	Ioannis		A-CASPER_WEB_D	DATA_SUBMITT -		Select		-		Search						Clear
3			Select A-CASPER_WEB_0	DATA_SUBMITTER		Select DC2										
Number of Entries found: 1			A-CASPER_WEB_1 A-CASPER_WEB_0	FECHNICAL_USER USER_ADMIN										Export CSV		Reset Vie
USER ID = USER NAME	≡ USER	ROLE	-	DATA COLLECTION CODE		COUNTRY	-	ORGANISATION	-	ASSIGNMENT	-	ACCESS RESTRICTION	-	# ASSIGNED ENTITIES	-	ACTIO
T	T		т		τ		т		τ		τ		τ		т	

Figure 30: Search panel details

The type of field depends on the type of data displayed – it may be a date field or a drop-down menu. All fields in the search panel marked with an asterisk (\*) are mandatory. To filter a list in the search panel, follow these steps.

- 1. Fill in the mandatory search fields. In the example shown in Figure 30, you can filter the list entries by choosing a username or selecting a user role or a specific data collection.
- 2. Click on "Search".
- 3. The number of entities found is displayed above the list.
- 4. The results are displayed in the list underneath.
- 5. Click on "Clear all" to remove all the filters applied. The list will go back to its default status.
- 6. The search panel is collapsible. You can close it by clicking on the small arrow on the left-hand side. Click on the arrow a second time to expand it.

### 1.4.5 Check/uncheck behaviour

When a list entry has a checkbox next to it, this means you can perform certain actions. To do this, first select the list entry by ticking the checkbox.

mt	er of Entries found	996														F	Download Files		Export CSV	R	eset Viev
	FILENAME	=	MODULE	-	REPORTING	=	ENTITY	=	ENTITY NAME	=	REFERENCE DATE	=	REMITTANCE DATE	=	RECEPTION DATE	1.=	ORIG./PARENT FILENAME	-	PROCESSING	=	ACTIO
		т		т		۲		T		۲	O DD / MM/ YYYY	۲	O DD / MIV/ YYYY	۲		т		T		۲	
]				_		-													Completed		Ĩ
1			_			_	_	-		-		-		-				-	Completed		1
		_				_				_				_		_			Completed		

#### Figure 31: Unchecked list entries

Figure 31 shows unchecked rows in a list. The blue button above the table is disabled, which indicates that no list entries have been selected for an action.

umb	er of Entries found	5: 996															Download Files		Export CSV	R	eset Viev
	FILENAME	=	MODULE	=	REPORTING	=	ENTITY TYPE	=	ENTITY NAME	=	REFERENCE DATE	=	REMITTANCE DATE	=	RECEPTION DATE	1=	ORIG./PARENT FILENAME	=	PROCESSING STATUS	=	ACTION
		۲		T		۲		Ŧ		۲	O DD / MN/ YYYY	۲	O DD / MM/ YYYY	T		۲		т		۲	
2				-		-				-		-							Completed		I
]		-				-													Completed		1
2																			Completed		T.

#### Figure 32: Checked list entries

When you select an entry from the list, the blue button is enabled, as displayed in Figure 32. To select list entries and perform an action, follow the steps below.

- 1. Click on the checkboxes to the left of the rows. A blue tick will appear, indicating that the list entry has been selected.
- 2. You can perform an action for all checked rows together. For example, if you wish to download both the files selected, simply click on "Download files".
- 3. To deselect the selected list entries, simply click on the checkboxes again and the tick will disappear.

### 1.4.6 Actions

You can perform certain actions for every row entry in a list through the CASPER Actions menu feature.

You can access the Actions menu for each row in the list by clicking on the "Actions" item in the row, as shown in Figure 33 or by right clicking anywhere on the row in question.

CODE	■ NAME	=	DESCRIPTION	=	OWNER	=	VALID FROM =	VALID TO =	DEFINITION STATUS	=	ACTIONS
	τ	۲		۲		۲	OD / MM / YYYY T	OD / MM / YYYY T		۲	-
-	-		UAT Mar 2019		ECB		01/03/2019		0		1
2TSTPE	- 1994						25/03/2019		View		2
ААА	AAA		AAA		SELVESA		15/03/2019		Edit Delete		
ABC	ABC		ABC		KH2		28/03/2019		0		-
в	в		asd		Asd		19/03/2019		0		1
BAA	BAA		BAA		BAA		02/05/2019		0		1

Figure 33: Example of the "Actions" menu

When you click on a row, a pop-up window will appear showing a list of actions and you can choose the action you want to perform for the row entry.

The items in the Actions menu are established according to the task and user permissions, as explained in Section 2. This means there may be different action items for each list. Action items may also vary for different users, depending on their permissions. For example, items may be disabled for one user role but enabled for others.

When an item is disabled, it is grey in colour and a tooltip will appear when you hover over it, as shown in Figure 34.



Figure 34: Disabled Actions item

### **1.5 Export functionality**

The "Export CSV" button displayed in CASPER is used to export the contents of a list. More precisely, it exports the current as-is status of the list, including the sorting, filters applied, table headers and any toggle columns that have been added to the view. The contents that are displayed are downloaded in a list (CSV) format. Please note that submission files are retrieved differently. The contents of a list can only be downloaded in CSV format. As shown in Figure 35, the downloaded file appears in the lower left-hand corner of your browser.

CSV stands for comma separated values, meaning that columns are separated by commas in the file.

Number of E	ntries fou	nd: 60								Add	Export		Reset Vie	N
CODE		NAME	=	DESCRIPTION	=	OWNER	=		VALID TO		DEFINITION STATUS	= ^	CTIONS	
	т		۲		۲		۲	OD / MM / YYYY T	OD / MM / YYYY	۲		T		
-	in and			UAT Mar 2019		ECB		01/03/2019			0		1	
-	-		-	-	-			25/03/2019			•		1	
AAA		-	-		6	-	-				•		1	
ABC		ABC		ABC		KH2		28/03/2019			0		1	
в		в		asd		Asd		19/03/2019			0		1	



### **2 HOW TO ACCESS CASPER**

CASPER is integrated into the ECB Identity Portal, the access management system for web users, so the platform can be accessed over the internet. In Section 2.1 you will learn more about the authentication process for CASPER, and in Section 2.2 about user roles and permissions.

### 2.1 Login (authentication process)

CASPER web users need to have an account on the ECB Identity Portal to be able to log in over the internet. Users who have this can access the platform using the following URL, as shown in Figure 36.

### https://id.ecb.europa.eu/login/

٢	EUROPEAN CENTRAL BANK   EUROSY	STEM
	Log in to ECB applications	
	Email Address *	
	Password *	
	✓ Log in	
	Frequently Asked Questions	

Figure 36: ECB Identity Portal login screen

Users on the ECB Identity Portal are grouped into organisations (also known as "levels"). These are legal entities, such as credit institutions or private

companies, that can manage their users independently. All accounts on the ECB Identity Portal must be assigned to the proper organisation. There are two types of web user in CASPER, depending on the organisation:

- 1. users belonging to specific organisations: these are managed by delegated access administrators (DAAs) within the same organisation;
- standalone users: these do not belong to any existing organisation on the ECB Identity Portal and submit data individually through CASPER (standalone users are assigned to a dummy organisation on the ECB Identity Portal named "ASTRA ECB").

CASPER roles are assigned by a CASPER access administrator based on this distinction, as set out below.

- If you belong to an organisation on the ECB Identity Portal, you need to contact either a DAA in your organisation or the ECB collection owner, who will grant you the correct CASPER roles. If you do not know who the CASPER DAA is in your organisation, please contact the ECB collection owner for assistance.
- 2. If you are a standalone user, you should contact the ECB collection owner, who will grant you the necessary CASPER roles to perform your tasks on the platform.

Once your roles have been granted, the CASPER tile will be displayed on the "My applications" page of the ECB Identity Portal, as shown in Figure 37.



Figure 37: ECB Identity Portal landing page

### 2.2 User roles and permissions

There are two user roles in CASPER over the internet: web user administrator and web data submitter. Each user has at least one role, and the different roles have different permissions.

As explained in Section 2.1, CASPER roles determine which tasks a user can perform. Access to data is controlled by the user administrators. To perform their tasks in CASPER, users need to be assigned (a) to a specific data collection, and (b) to the relevant entities.

For a comprehensive description of the different tasks performed by web users in CASPER, please refer to the individual user journeys described in Chapter 3.

This section describes the two web user roles: web user administrator and web data submitter. In the following paragraphs, you will learn more about web user

permissions and the responsibilities of the web roles in the data collection process.

### 2.2.1 Web user administrator

Web user administrators are responsible for managing access to data collections in CASPER for users who belong to the same organisation.

They have the right to grant or remove access for other web users in their organisation to any data collection assigned to them in CASPER, as described in Section 3.1. Web user administrators are also responsible for assigning web data submitters and other web user administrators in the same organisation to reporting entities.

To do this, a web user administrator must be assigned to the data collection and reporting entities by the ECB collection owner.

Web user administrators cannot prepare data collections, submit data or manage their own user assignment.

Web user administrators are responsible for managing the user assignment of web data submitters to the relevant data collections. They can also assign/unassign data submitters to reporting entities.

### 2.2.2 Web data submitters

Web data submitters upload reporting templates to CASPER to fulfil reporting obligations (for a definition of "reporting obligation", see the Abbreviations and terminology section). They can submit data only to the data collections and reporting entities to which they are assigned.

Data submitters are responsible for submitting data for the data collections and entities to which they are assigned.

### **3 USER JOURNEYS**

This chapter describes the user journeys of the different user roles in CASPER. In these journeys you will learn about the relevant functionalities, including access rights, screens and workflows. Each user journey consists of a series of steps that represent a typical scenario in which users interact with the system.

For more information on user groups, see Section 2.

### 3.1 Journey I: web user administrator

This journey describes the user management performed by web user administrators. Since CASPER release 1.2, the ECB has collected data from non-E(S)CB users such as commercial banks, private companies, universities and individual users who do not belong to a specific organisation. Web user administrators are responsible for managing access to their organisation's data. Please note that, depending on your role, you might notice some differences in the CASPER interface compared with the figures in this chapter.

Web user administrators can only manage the assignment of other users in the same organisation; they cannot manage their own user assignment. To be assigned to a collection, web user administrators need to contact either a collection administrator or another web user administrator for the collection in their organisation.

Web user administrators have the power to grant or remove access to any data collection to which they are assigned for users within the same organisation.

Figure 38 shows an overview of the individual steps in this user journey.



Figure 38: Overview of the web user administrator journey

### 3.1.1 User role request, approval and assignment

User access to CASPER is granted in a two-stage process.

1. User role management

2. Assignment of data collection and reporting entities in CASPER

CASPER is integrated in the ECB Identity Portal, the access management system for web users. If you have an account on the ECB Identity Portal, a CASPER role can be assigned to you. To use CASPER as a web user administrator, you need to request to be allocated WEB\_USER ADMIN status by contacting your CASPER access administrator. If you do not know who your CASPER access administrator is, please contact the ECB collection owner. For further information on web role assignments, see Section 2.1. You need an ECB Identity Portal account to access CASPER over the internet. Once you have this, you can log in using the URL below, as shown in Figure 39.

### https://id.ecb.europa.eu/login/

<b>E</b>	EUROPEAN CENTRAL BANK   EUROSY	STEM
	Log in to ECB applications	
	Email Address *	
	Password*	
	✓ Log in	
	Reset password Frequently Asked Questions	

Figure 39: ECB Identity Portal login screen

Once the role has been assigned, you will be able to access CASPER.

### 3.1.2 Request assignment to a collection

Once you have been granted a web user administrator role, you will be able to log into CASPER. However, you will not be able to assign users to collections.

First you need to send an email to the collection administrator requesting assignment to the collection and reporting entities you require.

### 3.1.3 Assign users to collections

As a web user administrator, you are responsible for assigning data submitters to data collections. You have to assign them to their data collections before they can access these.

CASPER provides simple steps for assigning a user to a data collection, as shown in Figure 40.

You can only assign users to data collections that have been assigned to you by the collection administrator or another user administrator.



Figure 40: User & Roles screen

- 1. Click on the "User & Roles" tab.
- 2. In the left-hand pane, click on "Users" to view the data collection assignment list. You can search by user ID, username, user role or data collection.
- 3. Enter your search criteria and click on "Search". The results are displayed in the grid below.
- 4. Select the user ID and the user's role in the grid, right click on the line and select "Assign".

A new window will open and you will be able to assign the user to one or more data collections, as shown in Figure 41. The list of collections available in this window depends on the collections to which you have been assigned.

Users	User ID	User Name	Country		User Role		
Roles			DE		APPR_CASPER_EXT_		
	DATA COLLECTI	ONS					Clea
		= CODE T-		= N	AME	=	
		mycol		T		т	
		MYCOL		h	AVCOLLECTION		Export C
	4					•	1

Figure 41: Data collection assignment screen

To assign the user to a collection, select the collection from the list and click on the checkbox, then click on the "Save" button.

### 3.1.4 Unassign users from collections

When necessary, you can also unassign users from a collection that you manage. This might be the case, for example, if a user transfers to a different business area within an organisation and submits data for another collection. As a user administrator, you are responsible for monitoring these matters and updating assignments to collections accordingly. Figure 42 and Figure 43 provide a step-by-step illustration of this process.



Figure 42: Data collection unassignment screen

• opuat	ed suce	cessfully!											
Active Set	arch												
USER ID			USER NAM	IE	USER ROLE			DATA COLLE	CTIONS	5			
					Select		•	Select		-		Search	
umber of Enti	ior four	vd: 42											
variabler of citie	ies ioui	10. 112				DATA							
JSER ID	=	USER NAME	=	USER ROLE	=	COLLECTION	=	COUNTRY	=	ORGANISATION	=	ASSIGNMENT	-
							τ.		<b>T</b>		T		

Figure 43: Data collection unassignment confirmation

Follow the steps below to unassign users from a collection.

- 1. Click on the "User & Roles" tab.
- 2. In the left-hand pane click on "Users" to view the data collection assignment list. Search by user ID, username or other fields.
- 3. Enter the user ID or name of the user that you would like to unassign from a collection and click on "Search". The data collections the user is assigned to are displayed in the grid.
- 4. Select the data collection in the grid, right click anywhere in the row and select "Unassign". To confirm your selection, click on "Confirm" in the popup window.
- 5. You will see the notification message: "Updated successfully". You have successfully removed the user from the data collection.

You can only unassign users from data collections that have been assigned to you. Assignments to other data collections are indicated by the entry "(Other DC)" in the data collection column. These need to be managed by the external user administrator for the collection concerned.

When a user role has been removed from the ECB Identity Portal, the Assignment column will display the entry "Role deleted".

### 3.1.5 Manage assignment to reporting entities

Once users have been assigned to a data collection and reporting entities created by the collection owner, user access to the entities can be established. Once users have obtained access to an entity, they will have access to the data for that entity. You are responsible for managing the access of other web users to entities, meaning that you can assign each individual entity (and data) to each user. See Section 3.1.5.1.

### 3.1.5.1 User assignment to entities

As a web user administrator, you can manage the assignment of entities to web users belonging to your organisation. Please note that this applies only if a user administrator has already assigned you to the data collection and entities concerned.

### 3.1.5.1.1 View entity assignment

The following steps explain how to view assigned entities. As a web user administrator, you can only view the entities to which you have been assigned.

			EURO	OPEA	N CENTRAL	BANK	EUR	OSI	STEM				~	Environ	
<b>A</b>			Cer	trali	ed Submissio	n Platfo	rm (C	ASP	ER)				_1 *	Logout	EN
						User & Rol	les								
66 M.	User Da	ta Collo	ction Ar	ciar	montlict										
_cc_wu	User - Da	ta Colle	cuon As	sigi	iment List										
															3
2	<ul> <li>Active Set</li> </ul>	arch													
gnment	USER ID		USER NAM	E	USER ROL	E		DATA	COLLECTIONS						
Access	cc_wds				Select	-		100	• • 005		Search		0	lear Al	П
															_
	Number of Entr	ries found: 1							Co	py Ema	ail Exp	ort CS	V Reset	t View	
	USER ID	■ USER	NAME	= U	SER ROLE	DATA COLLECT	ION	=	COUNTRY	=	ORGANISATIC	N ≡	ASSIGNMENT	=	AC RE
		T		<b>.</b>	T	CODE		т		т		т		•	KE:
		•		· _											
	casper_cc_wd	is@t Wds	Cc	C	ASPER_WEB_DAT	100005					CASPER Cont	rol T	Yes (Other DC	)	En
							Assig								
							Unas Edit E	-	Assignments						
									0	_					
							View	Entity	Assignments						

Figure 44: Data collection assignment list

User ID casper_cc_wds@ta	User Name	Organisation	ol Team	Organisation ID CASPER_Control	
User Role	1103.00	Data Collection Code	Data Collectio		
CASPER_WEB_DATA_SUE	BMITTER	100005	DC	n Name	
ENTITIES					
Number of Entries found: 5					Export CSV
REPORTING ENTITY					
COLLECTION UNIQUE ID	=	ENTITY NAME	=	REPORTING CODE	=
	т		т		т
RE01				RE01	
RE02		-		RE02	
RE03				RE03	
RE04		-		RE04	
RE07		1000		RE07	

Figure 45: View user assigned entities

- 1. Click on the "User & Roles" tab.
- 2. In the left-hand pane, click on "User assignment".
- 3. In the "Active search" pane, search for the user ID and data collection code for the user you want to assign to a reporting entity, then click on "Search".
- 4. Right click on the desired user and select "View entity assignment".

### 3.1.5.1.2 Edit entity assignment to users

As a web user administrator, you can assign and unassign users to the entities assigned to you.

The following steps explain how to assign users, typically web data submitters or other web user administrators in your organisation, to entities. Figure 46 and Figure 47 show the steps that must be followed.

				n Platform	(			· · · · ·
				User & Roles				
_	User - Data Col	llection Assignment List						
Jsers 2.	O Updated succes	sfully!					3	
User Entity Access	Active Search							
Report	USER ID	USER NAME	USER ROLE		DATA COLLECTIONS			
		Republication	Select	•	Select 👻	Search		Clear
	Number of Entries found:	3					Export CSV	/ Reset Vie
	USER ID = U	ISER NAME = USER ROLE =		COUNTRY	ORGANISATION	ASSIGNMENT	ACCESS	# ASSIGNED ENTITIES
	τ	Assign Unassign		τ	т т	τ.		т
	Annual scales [	Edit Entity Assignment View Entity Assignmen			CASPER	Yes	Entity	0

#### Figure 46: Data collection assignment list

User ID	User Name		Organisation			Organisation	ID
Sec. 1	second descendence		an Annual Second			tore.	
User Role			Data Collection Code		Data Collection Name		
CASPER_WEB_D	IATA_SUBMITTER		Strength Strength St.		Research to the Res		
NTITIES							
Number of Entries fo	ound: 6						Export CSV
	REPORTING ENTITY						
=	COLLECTION UNIQUE ID	= E	NTITY NAME	REI	PORTING CODE		-
		T					т
	BE	N	lationale Bank van Belgiū / Banque Nationale de Belgique	BE	E		
	DE	D	eutsche Bundesbank	DE	E		
	R	s	uomen Pankki	FI			
	GR	В	ank of Greece	GR	R		
	п	в	anca d'Italia	IT			
	PT	В	anco de Portugal	PT			5

Figure 47: Edit user assigned entities

- 1. Click on the "User & Roles" tab.
- 2. In the left-hand pane, click on "User assignment".
- 3. In the "Active search" pane, search for the user ID and data collection code for the user you want to assign to a reporting entity, then click on "Search".
- 4. Right click on the desired user and select "Edit entity assignment".
- 5. Select or unselect the desired entities from the list and click on "Save".

Note that you cannot assign a user to a reporting entity if their organisation ID differs from the reporting entity's ID. The only exception to this rule is when the organisation ID of the entity is left empty.

						E			NTRAL BANK   EU								1	<b>0</b> 2	Environme	EN I
					Submissions									User	& Roles					
hadran.	User Ent	ity A	ccess Repo	ort																3
Users 👻	Active S	earch																		
User Assignment 2	USER ID				USER NAME		USER ROLI						DATA COLI	ECT	IONS					
User Entity Access							Select All					~	DC2 ×							Ŷ
	ENTITY NAM	1E			ACCESS GROUP NAME															
																	Sea	irch	Clear	4
	Number of Ent	ries foi	ind: 505														Export CSV		Reset Viev	2
	USER ID	-	USER NAME	-	USER ROLE	•	COUNTRY	-	ORGANISATION		DATA COLLECTION CODE	•	ACCESS GROUP NAME	•	REPORTING CODE	•	ENTITY NAME	ACT	IVE	•
		۲		۲		۳		۲		۲		۲		۳		۳	· · · ·	· 🗆		τ
			1.1.1.1		A-CASPER_WEB_USER_ADMIN				ASTRA_ECB		DC2				ENTITY1		ENTITY1	Yes		Î
					A-CASPER_WEB_USER_ADMIN				ASTRA_ECB		DC2				ENTITY10		ENTITY10	Yes		
					A-CASPER_WEB_USER_ADMIN				ASTRA_ECB		DC2				ENTITY100		ENTITY100	Yes		
					A-CASPER_WEB_USER_ADMIN				ASTRA_ECB		DC2				ENTITY101		ENTITY101	Yes		

### Figure 48: User entity access report

- 1. Click on the "User & Roles" tab.
- 2. In the left-hand pane, click on "User entity access report".
- 3. In the "Active search" pane, insert the desired filters and press "Search".

### 3.1.6 View user entity access report

As a web user administrator, you can also see a summary of all users in your organisation assigned to given entities. Figure 48 shows the steps needed to perform a search in the user entity access report. You can search for different combinations of data collection, entities and user.

The user entity access report screen allows web user administrators to check users and their assignment to entities.

As a web user administrator, you have access to the data collections to which you are assigned. Within those data collections, you have access only to the entities that have already been assigned to you. Web user administrators are only authorised to see users in their own organisation.

### 3.2 Journey II: web data submitter

This journey describes the process for submitting files to CASPER. This is performed by data submitters.

Web users such as commercial banks, universities and individual users who do not belong to a specific organisation submit data over the internet. These users are called web data submitters.

Web data submitters are responsible for submitting data for the data collections to which they are assigned, and only on behalf of the reporting entities to which they are assigned.

In this section, you will learn about all possible scenarios that might occur during the data submission process. Please note that depending on your role you might notice some differences in the CASPER interface as compared with the figures in this chapter.

Web data submitters can only access a data collection to submit data if they have been assigned to it by the user administrator for that collection (see Section 3.1). They cannot assign themselves to a data collection.

Data submitters can access the list of reporting obligations and submit data for the entity to which they are assigned.

Figure 49 shows an overview of the individual steps in this user journey.



Figure 49: Overview of the data submitter journey

### 3.2.1 User role request, approval and assignment

User access to CASPER is granted in a two-stage process.

### 1. User role management

2. Data collection and assignment to reporting entities in CASPER

CASPER is integrated in the ECB Identity Portal, the access management system for web users. If you have an account on the ECB Identity Portal, a CASPER role can be assigned to you. To use CASPER as a web data submitter, you need to request a WEB\_DATA SUBMITTER role by contacting your CASPER access administrator. If you do not know who your CASPER access administrator is, please contact the ECB collection owner. For further information on web role assignments, see Section 2.1.

You need an ECB Identity Portal account to access CASPER over the internet. Once you have this, you can log in using the URL shown in Figure 50.

### https://id.ecb.europa.eu/login/

EUROPEAN CENTRAL BANK   EUROSYSTEM
Log in to ECB applications
Email Address *
Password *
<ul> <li>Log in</li> </ul>
<u>Reset password</u> <u>Frequently Asked Questions</u>

Figure 50: ECB Identity Portal login screen

Once the role has been assigned, you will be able to access CASPER.

### 3.2.2 Check notifications

CASPER generates notifications to inform users about established events in the collection lifecycle. By default, data submitters receive notifications for acceptance or rejection of submissions and other relevant events (the exact set of notifications may differ, depending on the setup for the data collection concerned). To view your notifications, log into CASPER and click on the mail symbol in the upper right-hand corner of the screen (Figure 51). The number on the notification icon shows new notifications that have not been read. Click on the icon to go to the notifications screen.

	EUROPEAN CENT	RAL BANK   EUROSYSTEM	Environment: Tr
	Centralised Submi	ssion Platform (CASPER)	🥗 🚽 🛔 Logout 🔳
۰ ۲	ubmissions		User & Roles
ANNOUNCEMENTS			Privacy Statement
TITLE01 (GENERAL)	FUTURE (GENERAL)	EIUHFKJDSAHFKDSJG (GENERAL)	SLKSGFDGFD FDSBVSFD
Announcement for automation created.	h	sfd	
Click here	Click here	Click here	
18 (GENERAL)	FZSDSF (GENERAL)	WELCOME TO CASPE (GENERAL)	R
18 sdr drfed casper.escb.eu/casper/#/announcementsl	http/t- dfgfg st		CASPER is up an running! We are experience! Please feel free to
Click here	S Click here	Click here	
			*

### Figure 51: CASPER landing page

The highlighted box shows the notifications icon. Click on the icon to open the notifications screen.

Figure 52 shows the notifications screen. The list on the screen displays all available notifications for the CASPER roles and data collections assigned to you.

				_
DATA COLLECTION	d: 9 DATE 2 © DD / MM / YYYY T	CATEGORY 3	Message	Export CS/ SEVERITY 5
MYCOL	25/04/2019 10:25:34	Submission	The Reporting Cycle WEEKLY of Data Collection MYCOL has started.	•
MYCOL2	24/04/2019 18:34:32	Submission	The Reporting Cycle MONTH of Data Collection MYCOL2 has started.	0
MYCOL	24/04/2019 10:06:59	Submission	The processing (or reprocessing) of submission C_MYCOL_DAILY_MYMOD1_MYENTITY1	0
MYCOL	23/04/2019 15:22:05	Submission	The processing (or reprocessing) of submission C_MYCOL_DAILY_MYMOD1_MYENTITY1	0
MYCOL	23/04/2019 15:19:38	Submission	The processing (or reprocessing) of submission C_MYCOL_DAILY_MYMOD1_MYENTITY1	0
MYCOL	23/04/2019 13:05:00	Submission	The processing (or reprocessing) of submission TC_MYCOL_DAILY_MYMOD1_MYENTITY	0
MYCOL	18/04/2019 13:37:34	Submission	Submission C_MYCOL_DAILY_MYMOD1_MYENTITY1_20190416_1_GR.xisx has been rejec	0
MYCOL	17/04/2019 15:36:20	Submission	The processing (or reprocessing) of submission C_MYCOL_DAILY_MYMOD1_MYENTITY1	0
MYCOL	16/04/2019 16:10:57	Submission	The Reporting Cycle DAILY of Data Collection MYCOL has started.	•

### Figure 52: Notifications screen

Each entry in the list represents one notification and contains the information shown in the table below.

- 1. Data collection to which the notification relates
- 2. Exact date and time of notification
- 3. Category of notification
- 4. Description of notification
- 5. Severity of notification

There are three different kinds of notification severity in CASPER. A green icon indicates that an action has been completed successfully and no further action is required from the user. Amber indicates that the action has been completed but with a warning. Red indicates that a fatal error has occurred and requires user action.

Notifications are generated for specific CASPER events. There are nine events which could trigger a notification for a data submitter. Table 5 describes each event in detail.

Event type	Description
Submission accepted and a reporting obligation fulfilled	A successful file submission has been performed. Both the technical and business validation processes have been completed with no fatal errors. You still need to look at the result of the business validation to check for non-fatal errors and warnings.
Submission overdue	The deadline for a file submission has already expired. Submit the file as soon as possible.
Submission rejected (technical validation failed)	A file submission has failed because the technical validation process encountered an error. You will need to correct the error and resubmit, as the reporting obligation has not been fulfilled.
Submission rejected (business validation failed)	A file submission has failed because the business rules evaluation process encountered a fatal error. You will need to correct the error and resubmit, as the reporting obligation has not been fulfilled.
Reporting obligation created	A reporting obligation for your organisation has been created. You can submit data once the reporting cycle starts.
Reporting cycle started	The reporting cycle for a reporting obligation has started and you can now submit data.
Reporting cycle closed	Once the reporting cycle has closed, you will no longer be able to submit data. For more information on reporting cycles and data submission deadlines, see Section 3.2.5.

Event type	Description
Attachment upload succeeded	Notification confirming the successful upload of an attachment to a submission.
RA template upload succeeded	Notification confirming the successful upload of a reporting agent (RA) template.

Table 5: Data submitter notifications

### 3.2.3 View data collection information

As a data submitter, you can view detailed information about the data collection assigned to you.

		EUROPEAN CENTRAL BAN				Environme
	Su	bmissions	1		User & Roles	
ata Collections						
Number of Entries found: 1						Export CSV Reset View
CODE T	-	NAME	-	DESCRIPTION		
1	۲		τ			
		DC100005		Test	View Hi View Va View Di View Re	lidation Results scussions

Figure 53: Submission landing page

To do this, follow the steps shown in the table below.

- 1. Go to the submissions screen and select the data collection.
- 2. Click on "Actions" and then "View submission list".
- 3. On the submission list screen, click on "Information".

You can see the data collection information by clicking on "Information" on the left-hand side of the screen (see Figure 54).

									RAL BANK   EUROS						@ ≥ ≛	Crivironi ogout
				Submis	slons							User 8	Roles			
ho di se	Sub	miss	ion List													
Information Submission List	Num	ber of En	tries found: 25	o						Refresh		Download Files	Zip Upload	Export C	SV Reset	View
File Vault		• =	CYCLE	REPORTING	1.78	MODULE 2 *	ENTITY NAME		REFERENCE DATE	OBLIGATION STATUS	•	RECEPTION 3 VE	REMITTANCE	4 .≡	ATTACHMENT	АСТ
Validation Results		<b>T</b>		r	۲			۲	O DD / MM / YYYY T	·	۲	T	O DD / MM	/ mm <b>T</b>		
Discussions		0	RCAT8	CENTITY		ATMOD	centity		30/11/2021	0			30/11/2021			1
Reports		0	ATCYX	CENTITY		ATMOD	centity		15/10/2021	ø			15/10/2021			- 1
		~ 5	NEWV	CENTITY		ATMOD	centity		15/10/2021	A		28/01/2022 12:08:35	15/10/2021		8	- 1
		0	NEWV	CENTITY		ATUL2	centity		15/10/2021	0			15/10/2021			- 1
		0	HIIII	CENTITY		ATULMOD	centity		15/10/2021	0			15/10/2021			1
		0	ATULMO	CENTITY		ATULMOD	centity		15/10/2021	0			15/10/2021			- 1
		~ 2	ATUL8	CENTITY		ATULMOD	centity		26/10/2021	A		26/10/2021 10:03:59	26/10/2021			- 1
		1	NEWV	CENTITY		ATULMOD	centity		15/10/2021	A		15/10/2021 17:24:19	15/10/2021			1

### Figure 54: Submission list information

By clicking on "Information", a new pop-up screen is displayed with information about your data collection, i.e. the data collection code, the name and a description. You can also see an email address with the contact details of the data collection owner, if provided.

Information		
Code	Name	
MYCOLL	My Data Collection	
Description User manual data collection		
		Close

Figure 55: Submission List information pop-up

### 3.2.4 Event-based data collections

CASPER supports two types of data collection. Standard data collection is used when the collection owner expects a reporting agent to submit data with a predefined reference date. For a standard data collection, the reporting agent needs to fulfil a reporting obligation by performing a valid submission before the remittance date. In contrast, event-based data collection is used when the collection owner expects the reporting agent to submit data based on the occurrence of a certain event. In this case, the reference date of the submission is not known in advance. For event-based data collections, the reporting agent is expected to submit only if an event occurs. This is so that an obligation can be considered fulfilled with zero or multiple submissions.

### 3.2.5 Check reporting obligations

CASPER reporting obligations entail a reporting entity submitting a module file for a given reference date. In a traditional data collection, a reporting obligation is fulfilled if at least one completed submission complies with the reporting obligation for your organisation. In an event-based data collection, a reporting agent is supposed to submit only if a certain event occurs. This means that an obligation can have zero or multiple submissions depending on the number of events that have occurred. This section will illustrate how to check reporting obligations for standard and event-based data collections.

A collection owner is the person responsible for collecting data and monitoring the submissions for a data collection.

As a data submitter, you can view the reporting obligations for your organisation in data collections assigned to your user and submit data. To view the reporting obligations for a collection assigned to you, click on "Submissions" in the top menu bar, as shown in Figure 56.

- 1. The submissions screen opens. This displays a list of the data collections assigned to you, including the code, the name and a description for each one.
- 2. Select a data collection and click on "Actions". A small pop-up box will appear with four options: you can go to the submission list screen, the file vault, the validation results or the discussion screen.
- 3. To view reporting obligations, click on "View submission list". This will take you to the submission list screen.

The submission list screen displays all reporting obligations for a given collection for which you can submit or have already submitted files. Note that the submission list will not display information for files that have failed a technical validation or business validation with a fatal error. To view all the files you have submitted regardless of the final processing result, go to the file vault screen as described in Section 3.2.11.

		Submissions			User & Roles		
			<u> </u>				
ata Collections							
Number of Entries found: 2					Export	csv	Reset View
CODE T	=	NAME	= DE	ESCRIPTION		=	ACTIONS
spr	۲		Ŧ			۲	
12SPR							1
120111					View	Submission	List
						File Vault Validation F	

#### Figure 56: Submissions screen

For standard data collection, Figure 57 shows the submission list screen. A table is displayed with all the reporting obligations that belong to this data collection and your organisation.

					EUROPEA	N CENTRAL	BAN	K   EUROSYSTE	м						Environm	
<u> </u>					Centrali	sed Submissi	on Plat	form (CASPER)						<b>2</b> • •	Logout	E١
				Submissions								User & Roles				
	Sub	miss	ion List													
Submission List	Num	ber of En	ries found: 6						Dov	vnload Files		Zip Upload	Export	CSV Reset	View	כ
File vault Validation Results		. =	CYCLE T =	REPORTING 1 *= CODE	MODULE 2	ENTITY NAM	4E ≡	REFERENCE DATE	=	OBLIGATION STATUS	=	RECEPTION DATE	3 •≡	REMITTANCE DATE	4 .≡	
Discussions		۲	cycle1 T	T		τ	۲	O DD / MN/ YYYY	۲		T			DD / MN / YYYY	٣	
Reports		~ 2	CYCLE1	AT	MOD1	Entity1		04/05/2020		0		(Select All)		3/05/2020		
		~ 6	CYCLE1	AT	MOD1	Entity1		27/04/2020		A	~	Not Received Not Received	- Not Due	Yet)/04/2020		
		× 2	CYCLE1	BE	MOD1	Entity 2		27/04/2020		A		Not Received Received - Acceived	epted	)/04/2020		
		~ 1	CYCLE1	BE	MOD1	Entity 2		04/05/2020		0		A Received - En		3/05/2020		
		× 1	CYCLE1	E3	MOD1	Entity 3		27/04/2020		A		A Received - Wa		)/04/2020		
		~ 1	CYCLE1	E3	MOD1	Entity 3		04/05/2020		A	V	A Revalidated - Revalidated - Revalidated - Revalidated -	Fatal Valid	3/05/2020		

Figure 57: Submission list screen showing reporting obligations

The submission list screen and file vault screen are the main screens for data submitters: they are therefore referred to frequently in the later sections of this journey.

Table 6 gives an overview of all submission list attributes in the default view.

Attribute	Explanation
#	The column displays the total number of resubmissions. For child entries on the submission list screen, this field is blank. If there is no submission, it shows "0".
Cycle	The reporting cycle is the time frame within which submissions of a specific set of modules for a given reporting obligation are accepted by CASPER. In the example above (Figure 57), the reporting cycle is named "CYCLE1".
Reporting code	The code which identifies the reporting entity.
Module	The code of the module as established by the collection owner:

Attribute	Explanation
	<ul> <li>if it requires signing, the following icon will be displayed</li> </ul>
	next to the module name: 🏶 ;
	<ul> <li>if it requires encryption, the following icon will be displayed</li> </ul>
	next to the module name: $lacksquare$ .
Entity name	The name of the reporting entity.
Reference date	The date to which the reported data refer.
Obligation status	The obligation status (Table 8).
Reception date	The date when the system received the file.
Remittance date	The due date for file submission.
Attachment	Indicates whether an attachment to the submission has been uploaded. Possible values:
	🔇 (yes);
	(no).
Actions	The list of actions for the reporting obligation.

Table 6: Submission List default attributes

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table 7.

Attribute	Explanation
Delay in days	<ul> <li>The number of days between:</li> <li>the remittance date and the date of receipt of the first successful submission;</li> </ul>

Attribute	Explanation
	<ul> <li>the remittance date and the current date, if no submission has been made.</li> </ul>
Failed attempts	The number of failed attempts before a successful submission.
Sequential number	The sequential number of the file (as explained in Section 3.2.9.2).
Entity type	The type of reporting entity, i.e. company, group or person.
Entity group	The name of the reporting entity group, as established by the collection owner.
Country	The country of the reporting entity (not applicable).
Cycle status	The status of the reporting cycle. Possible values:
	• started;
	• closed;
	• restarted.
Module version	The version of the module.
Expected	Indicates whether the reporting obligation is mandatory. Possible values:
	• Y (yes);
	• N (no).

Attribute	Explanation
Validation status	Status of the file after applying the business validation rules. Depending on the severity level of the highest rule failed, the status may be:
	• warning;
	• error;
	• fatal.
	If all rules are passed, the status will be:
	• valid.
	If the business validation rules were not executed, the status will be:
	• NA.
Revalidation date	Timestamp of the revalidation. Blank if the file has not been revalidated.
Number of errors	Number of validation rule failures that occurred during the business validation process with a severity level "error". Blank if no submission was received.
Number of warnings	Number of validation rule failures that occurred during the business validation process with a severity level "warning". Blank if no submission was received.
Export flag	Indicates that it has not been possible to export the submission data to DISC for technical reasons. Please contact a system administrator for assistance. Possible values:
	<ul> <li>Y (data have not been transferred);</li> </ul>
	• (blank).
Requires encryption	Indicates whether the module submissions are to be encrypted by the RA. Possible values:
	Yes

Attribute	Explanation
	• No
Requires signing	Indicates whether the module submissions are to be signed by the RA. Possible values:
	<ul><li>Yes</li><li>No</li></ul>

Table 7: Submission List additional attributesPlease note that you might not be able to see files submitted in the past for reporting cycles that are now closed. Collection owners can hide obsolete data in the submission list.

On the upper right-hand side of the screen there are three blue buttons. Details of their functions are provided below.

**Export CSV:** Use this button to export the list view in CSV format.

**Zip upload:** Use this button to submit multiple files in .zip format. For more information on zip uploads, see Section 3.2.9.2.

**Download files:** Use this button to download the file(s) you have submitted. First enable the button by selecting at least one item from the list, checking the box at the beginning of a row. You can also select multiple items, in which case all the files selected will be zipped and downloaded.

There are two ways of checking your upcoming reporting obligations. First, the "Remittance date" column displays the due date for submissions for the reporting obligation concerned. Second, the "Status" column shows specific information about the current status of the reporting obligation. If you hover your mouse cursor over the icon, you will see the status details.
Table 8 shows the different types of reporting obligation status as displayed in the submission list, with a description.

Obligation status	Explanation
Not received – delayed	The remittance deadline for a file submission has already expired. Submit the file as soon as possible.
Not received – not due yet	The remittance deadline for a file submission has not yet arrived.
Not received – optional	File not received. However, submission of a file for this reporting obligation is not mandatory.
Received – valid	File received and both technical and business validations have been passed with no errors or warnings.
Received – warning	File received. Technical validations have been passed but the business validations encountered at least one error with a severity level "warning". Your submission has nevertheless been accepted. You will need to contact the collection owner to check whether any further action is necessary.
Received – error	File received. Technical validations have been passed but the business validations encountered at least one error with a severity level "error". Your submission has nevertheless been accepted. You will need to contact the collection owner to check whether any further action is necessary.
Revalidated – valid	A submitted file has been revalidated by the collection owner (e.g. after deactivating a validation rule) and the business validations have been passed.
Revalidated – warning	A submitted file has been revalidated by the collection owner (e.g. after deactivating a validation rule) and the business validations encountered at least

Obligation status	Explanation
	one error with a severity level "warning". The revalidation has nevertheless been accepted. You will need to contact the collection owner to check whether any further action is necessary.
Revalidated – error	A submitted file has been revalidated by the collection owner (e.g. after deactivating a validation rule) and the business validations encountered at least one error of with a severity level "error". The revalidation has nevertheless been accepted. You will need to contact the collection owner to check whether any further action is necessary.
Revalidated – fatal	A submitted file has been revalidated by the collection owner (e.g. after deactivating a validation rule) and the business validations encountered at least one error with a severity level "fatal". Resubmission is expected in these cases.

Table 8: Obligation status types

# As a data submitter, by default you cannot revalidate a file submission yourself.

Collection owners establish a reporting obligation as either mandatory or optional. For mandatory obligations, the collection owner expects to receive a file submission fulfilling the reporting obligation concerned before the remittance deadline. However, optional reporting obligations grant the option to submit a file for the reporting obligation. The submission file status for an optional obligation will not change to "Not received – delayed" even if the remittance date has expired and no file has been submitted.

All optional obligations will automatically disappear from the submission list screen once the status of the reporting cycle is "closed" (i.e. after the end date of the cycle), even if you have already submitted the file(s). However, the files will remain visible on the file vault screen.

You can submit multiple files, called resubmissions, for the same reporting obligation. The entries for multiple submissions are added as sub-entries to the main reporting obligation entry. You always see the latest submission as the main reporting obligation entry. To see all submissions, simply click on the small arrow in the # column to expand (as displayed in Figure 58).

9														Download Fi	165	Zip Upload		Export CSV	Re	set View
	• =	CYCLE	-	REPORTING	1 *=	MODULE 2*	-	ENTITY NAME	-	REFERENCE DATE		OBLIGATION	-	RECEPTION 3 DATE	-=	REMITTANCE	4 -=	ATTACHMENT	=	ACTIONS
	۲		۲		۲		۲		۲	O DD / MM/ YYYY	۲		۲		۲	O DD / MM/ YYYY	٣		۲	
	× 1	CYC2		ENT1		MOD1		entity 1		01/04/2020		<b>A</b>		01/04/2020 18:12:5	i6	02/05/2020				1
	^ 34	CYC1		ENT1		MOD1		entity 1		02/03/2020		<b>A</b>		01/04/2020 17:22:5	i2	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		<b>A</b>		01/04/2020 17:22:5	2	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		A		01/04/2020 17:17:1	3	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		4		01/04/2020 17:17:1	3	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		0		01/04/2020 15:24:3	19	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		۲		01/04/2020 15:24:3	19	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		<b>A</b>		01/04/2020 15:18:5	14	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		۲		01/04/2020 15:17:2	19	05/04/2020				1
		CYC1		ENT1		MOD1		entity 1		02/03/2020		<b>A</b>		01/04/2020 15:13:3	18	05/04/2020				1

Figure 58: Parent-child relationship for resubmissions

In the case of event-based data collections, the "Submission List" will be as per Figure 59.

									ean centi alised Submi						🕸   🕿   😵						
<i>)</i>	Submissions							Collections		l I		Dictionaries	Dictionaries			User & Role					
ITST	Subm	ission Li -	st																		
ctions		able Event													_				_		
ission List ault ition Results	Numb	er of Entries fou	nd: 6 🖸	=	REPORTING CO	DE 1*	=	MODULE	2 *	=	FORMAT 3	•	=	Refresh ENTITY NAME		Hew Event	D NTITY TYP	port CSV	_	ACTIONS	
ssions eletion Report ts		ROUND1 TEST ROUND1			DEABCDEFGHI DEABCDEFGHI FRABCDEFGHI	123456789			EVENT1 EVENT1 EVENT1		Excel Excel Excel			Test Bank 1 Test Bank 1 Test Bank 3		G	roup			1	
		itted Eve																			
		er of Entries fou	LE	REPO		MODUL	E 2*	= T	ENTITY NAME	EVENT DAT	те =	STATUS T	-	RECEPTION		ATTACHMEN	-	port CSV	_	ACTIONS	
		1 ROU ~ 1 ROU			CDEFGH01234		EVENT		Test Bank 3 Test Bank 3	25/02/202		A		27/02/2023 10:0				•		1	

#### Figure 59: Submission list for event-based data collections

The screen is divided into two parts. The top part displays the available events for which you can submit your files. The second part displays the list of events for which you have already submitted files. Table 9 gives an overview of all the attributes in the default view.

Attribute	Explanation
Cycle	The reporting cycle is the time frame within which submissions of a specific set of modules for a given reporting obligation are accepted by CASPER.
Reporting code	The code that identifies the reporting entity.
Module	The code of the module as defined by the collection owner.
	If it needs to be signed, the following icon
	will appear next to the module name: 🏶

Attribute	Explanation
	If it requires encryption, the following icon
	will appear next to the module name: 🖴
Entity name	The name of the reporting entity.
Format	The format of the module, i.e. Excel, CSV or XML.
Entity type	The type of entity.
Actions	This displays the list of actions for the event, such as:
	<ul> <li>New event: perform a submission for an available event;</li> </ul>
	<ul> <li>Revalidate: trigger a revalidation (only possible for internal collection designers);</li> </ul>
	<ul> <li>Download template: download the template of the reporting obligation;</li> </ul>
	<ul> <li>View validation rules: visualise the validation rules implemented for the particular event.</li> </ul>

## Table 9: Submission list attributes for event-based data collections

In addition to the columns that are visible by default, you can add toggle columns for the attributes as shown in Table 10.

Attribute	Explanation
Cycle start date	Start date of the available event
Cycle end date	End date of the available event: after this date, an event will not be "available" anymore and will disappear from this list.

Attribute	Explanation
Entity group	Entity group of the reporting obligation
Cycle status	<ul> <li>This show the status of the cycle and can be one of the following values:</li> <li>Closed;</li> <li>In preparation;</li> <li>Restarted;</li> <li>Started.</li> </ul>
Module version	The version of the module
Export flag	<ul> <li>The export flag indicates that submission data could not be exported to DISC for technical reasons. Please contact a system administrator for assistance.</li> <li>Possible values are: <ul> <li>Y (data have not been transferred);</li> <li>(blank).</li> </ul> </li> </ul>
Requires encryption	This attribute means that the module submissions are to be encrypted by the reporting agent. Possible values are: • Yes; • No.
Requires signing	This attribute appears if the module submissions are to be signed by the reporting agent. Possible values are: • Yes; • No.

#### Table 10: Optional attributes for submitted events

The lower part of the screen in Figure 60displays the list of submitted events. A detailed explanation of the attributes that appear in the default view can be found in Table 11.

Attribute	Explanation
Cycle	The reporting cycle is the time frame within which submissions of a specific set of modules for a given reporting obligation are accepted by CASPER. In the example above (Figure 57), the reporting cycle is named "DAILY".
Reporting code	The code that identifies the reporting entity.
Module	The code of the module as defined by the collection owner.
	If it needs to be signed, the following icon
	will appear next to the module name: 🏶
	If it requires encryption, the following icon
	will appear next to the module name: 🔒
Entity name	This is the name of the reporting entity.
Event date	This is the event date of the submission. This information is filled in by the reporting agent whenever an event- driven submission is made.
Obligation status	This is the status of the obligation, as shown in Table 8.
Reception date	The date on which the file was received.
Attachment	This indicates whether an attachment to the submission has been uploaded. Possible values: • (yes); • (no).
Discussion	This is the status of the discussion for a particular submisson. If a discussion exists for a particular submission, this attribute will be populated by an orange

Attribute	Explanation
	chat icon and will otherwise be left empty.
Actions	This is the list of actions for a submission:
	Resubmit event: re-upload a file for that particular event;
	Download: download submitted file;
	<ul> <li>View validation results: visualise any failing validation rule;</li> </ul>
	<ul> <li>Download template: download the template of the reporting obligation;</li> </ul>
	<ul> <li>Revalidate: run a revalidation or the submitted file. This option is only available for collection owners;</li> </ul>
	Upload attachment: upload an attachment;
	<ul> <li>Download attachment: download the last attachment which was submitted;</li> </ul>
	<ul> <li>View validation rules: visualise the validation rules implemented for the particular event;</li> </ul>
	Discuss: start a discussion regarding the submission.

## Table 11: Submission events list

In addition to the columns that are visible by default, you can add toggle columns for the attributes as shown in Table 12.

Attribute	Explanation
Event comment	Comment entered by the data submitter during the submission process
Failed attempts	The number of failed attempts before a successful submission
Sequential number	The sequential number of the file (as explained in Section 3.2.9.2)
Entity type	The type of reporting entity i.e. company, group or person
Entity group	The name of the reporting entity group as defined by the collection owner
Entity Country	This is the country of the reporting entity.
	This column is populated only if the access restriction of the data collection is set to "country" (as per Section 3.2.5).
Cycle status	This is the status of the reporting cycle. Possible values are:
	• started;
	• closed;
	• restarted.
Validation status	Status of the file after applying the business validation rules. Depending on the severity level of the validation rules, the status may be:
	warning;
	• error;
	• fatal.
	If all rules are performed, the status will be:
	• valid.
	If the business validation rules were executed, the status will be:
	NA. : "Not applicable"

Attribute	Explanation
Revalidation date	This is the timestamp of the revalidation, and remains blank if the file has not been revalidated.
Number of errors	This shows the number of validation rule failures that occurred during the business validation process with the severity "error". This field is blank if no submission has been received.
Number of warnings	This shows the number of validation rule failures that occurred during the business validation process with the severity "warning". This field is blank if no submission was received.
Export flag	This indicates that submission data could not be exported to DISC for technical reasons. Please contact a system administrator for assistance. Possible values are:
	<ul> <li>Y (data have not been transferred);</li> <li>(blank)</li> </ul>
	• (blank).
Requires encryption	This attribute means that the module submissions are to be encrypted by the reporting agent. Possible values are:
	• Yes;
	• No.
Requires signing	This shows whether the module submissions are to be signed by the reporting agent. Possible values are:
	• Yes;
	• No.

Table 12: Submission events list additional attributes

## 3.2.6 Download reporting template

A file submitted to CASPER will only be accepted if it is based on the reporting template for the reporting obligation in question. The reporting template is a file configured by the collection owner that has to be completed by the data submitters and uploaded to CASPER. It has the requisite format and structure, and it contains either dummy data or no data at all. Please note that the collection owner might not provide the template.

For detailed instructions on how to complete a reporting template, see Section 3.2.8.

If the reporting template is supplied by the collection owner, data submitters can download it the from the submission list – the same view used to monitor reporting obligations and perform submissions. Figure 60 and Figure 61 show how to download a reporting template.

			N CENTRAL BANK	·	1	Envir
			Submissio			
Data Collections						
Number of Entries found: 2					Export CS	V Reset View
CODE	=	NAME	=	DESCRIPTION		= ACTIONS
	۲		т			· 🤈
And the second sec		DC		Test		1
Restored		hodical	Vi Vi Vi	ew Submission List ew File Vault ew Validation Results ew Discussions ew Reports		I

Figure 60: Submissions screen for downloading a reporting template

								E			ITRAL BANK   EUR						<u>o</u> .	Enviro
											Submissions							
to do at	Sub	mi	ssi	on Lis	st													
Submission List	Num	ber o	f Entri	ies found:	2									Download Files	Zip Upload	Đợ	port CSV	Reset View
File Vault Validation Results			-	CYCLE	=	REPORTING CODE	1 *=	MODULE 2 *=	ENTITY NAME	-	REFERENCE DATE	OBLIGATION STATUS	=	RECEPTION 3 -=	REMITTANCE	4	ATTACHMENT	
Discussions			٣		۲		۲	τ		۲	OD/MN/YYYY T		۳	τ	O DD / MW/ YYYY	۲	· · ·	3
Reports			0	RC1		RE01		MOD1	test		13/04/2021	0			14/04/2021			1
			0	RC1		RE01		MOD2	test		13/04/2021	0		Upload File Download File View Validation Download Ten Oproto Pictation Download Atta View Validation	n Results nplate	4		I

#### Figure 61: Downloading a reporting template

The steps shown in the table below should be followed.

- 1. Go to the submissions screen and select the data collection for which you would like to download the reporting template.
- 2. Click on "Actions" and then "View submission list".
- 3. On the submission list screen, select a reporting obligation.
- 4. Click on "Actions" and then "Download template".

## 3.2.7 Check business validation rules

One of CASPER's key features is that it validates the data submitted. If configured by the collection owner, the platform will validate uploaded files against all validation rules applicable to the reporting obligation. It is important for data submitters to understand the validation rules against which their data will be checked. CASPER therefore enables these rules to be viewed before data submission.

On the "Submission List" screen, select the reporting obligation (or the available event in the case of event-based data collections) for which you would like to

check the validation rules, then click "Actions" and select "View Validation Rules", as shown in Figure 63 below.

									E	UROPEAN C	EN	TRAL BANK	UR	DSYSTEM							Enviro
										Centralised	Sub	mission Platforn	(C)	ASPER)					2	-	Logout E
												Submissions									
0.00	Sub		sio	on List																	
ibmission List	Numb	ver of I	Entrie	is found: 2												Download Ries	Zip Upload	Exp	oort CSV	Re	set View
e Vault lidation Results				CYCLE	-	REPORTING	1 *=	MODULE	2 -=	ENTITY NAME	=	REFERENCE DATE	=	OBLIGATION	=	RECEPTION 3 -=	REMITTANCE	4 -≡	ATTACHME	NT	ACTIONS
scussions			•		Ŧ		۲		۲		۲	OD/MV/YYYY	۲		۲	T	O DD / MW/ YYYY	٣		۲	
ports		3		RC1		RE01		MOD1		test		13/04/2021		0							1
		1		RC1		RE01		MOD2		test		13/04/2021		0		Upload File Download File View Validatic Download Tei Upload Attact Download Att View Validatic	n Results nplate iment achment				I

Figure 62: Validation rules navigation

The system redirects you to the validation rules screen. Figure 63 shows the different areas of the screen. The highlighted box displays key attributes for the reporting obligation. The list shows all the validation rules applicable to the submission concerned. Table 13 describes the attributes of the validation rules.

			EUROPE	N CENTRAL BAN	K EUROSYS	TEM					-		Environmen
			Centrali	sed Submission Pla	tform (CASPE	R)					କ୍	-	Logout E
5		Submissions							User & Roles				
	Validation R	ules											
ubmission List 👻	Cycle			Module Code				Re	eporting Code				
Validation Rules	CSVCYC												
le Vault	Entity Name			Reference Date									
alidation Results iscussions	ent3												
eports	Number of Entries fo	und: 8								E	Export CSV	Re	set View
	RULE ID	FORMULA LEFT =	OPERATOR =	FORMULA RIGHT	INTERVAL	-	INTERVAL AS %	=	ERROR MESSAGE	=	SEVERITY	=	ACTIONS
	T		· · · · · · · · · · · · · · · · · · ·			۲		۲		۲		۲	
	12345	KEYS(DPS(COMPILI	-	"hi"	0		N		error		ERROR		1
	MATCH_CHARACT.	MATCH_CHARACTE	-	TRUE	0		N		error		View		

Figure 63: Validation rules screen

Attribute	Explanation
Rule ID	The unique ID of the validation rule.
Formula left side	The left-hand side of the formula, evaluated after the submission.
Operator	The comparison operator that compares the content of the left-hand side of the formula with the right-hand side.
Formula right side	The right-hand side of the formula, evaluated after submission.
Interval	The range for value deviation. This is a positive integer.
Interval as %	Indicates whether the interval is a percentage or an absolute value. Possible values:
	• Y (yes);
	• N (no).
Error message	The error message if validation fails. The message may include the specific failed observation value.
Error severity	The severity of the error. There are three levels of severity:
	• warning;
	• error;
	• fatal.
Description	A detailed description of the validation rule.
External rule	Possible values:
	• Y (yes) if the rule is external and therefore not executed in CASPER;
	• N (no) if the rule is internal.

Attribute	Explanation
Actions	<ul> <li>The list of actions you can perform for every file. Only one option is available:</li> <li>view the definition of the validation rule.</li> </ul>

Table 13: Validation rule attributes

You can view the definition of the validation rules. Select the rule and click on "Actions" and then "View". This opens a read-only pop-up window which shows important information about the definition of the rule, as described in Table 12. Figure 64 shows the pop-up window.

tion Ru ABasic Det	tails					1 Alexandre
Rule ID*			Description			
R04	D B	ternal Rule				
me Module Code	e*					
MYMODULE	E3					
of Entries fou						
0	🗌 In	iterval As %				
Severity*						IRITY
WARNING						
						WARNING
▲ Formula						
Formula Left	Side*	Formula Operati	or*	Formula Right Side*	0	
DPI(NUMBE	R2_OPT)	>=		4		- 100
Error Messag	ge Definition*					
Data point i	is less than 4. Please ch	eck NUMBER2 OPT				*

Figure 64: Validation rule pop-up window

Business validation rules follow the proprietary CASPER validation rule syntax, combining functions, operators and data points for data validation. Collection owners might therefore need to provide further instructions to data submitters to inform them precisely how the data will be validated.

## 3.2.8 Template completion rules

An essential step in the data collection process is to complete the reporting template by entering the relevant data according to the completion rules set by the collection owner. Mistakes during the data preparation process, or changes to the reporting template, could lead to unexpected validation results or even the rejection of the submission. Once you have downloaded a reporting template, it is important to understand the rules on how and where to enter the data. In some cases, templates include instructions on how and where to fill them in. If you require further information, please contact the collection owner.

Some key points to help you in this process are shown below.

Do not make changes to the structure of the template provided, such as adding or removing worksheets. Do not change worksheet names.

Do not alter the existing password protection, such as adding or removing the protection for the workbook or single worksheets.

Do not change the formatting of the template, such as changing the data type or number of decimal places (number type only) for cells that are being reported (observation cells).

Do not change the content of any cells other than observation cells, i.e. the cells for which data are required. Some templates contain Excel formulas to calculate certain cell values. Do not change any formulas.

Read the column headers as well. They explain what type of data are required for the observation cells.

A sample reporting template with observation cells is shown in Figure . The actual template may differ in terms of layout, column headers and completion instructions.

		Wrap Text General -		ter Impor Writer Im		v Goran a Fill v		
ter B I U - 🖂 -	<u>◇·▲</u> ·≡≡≡ # # # ⊞	Merge & Center - 📑 - % 🔹 % C Fo	Conditional Format as Writ	ter Impor Writer Im	por Writer Impor	· ▼ Insert Delete Format		
Fig Font	Fa Alignment	Fa Number Fa		Styles		Cells Editing		
• (* <i>f</i> x								
В	С	D	E	F	G	Н		
r fill-in instructions			9		ECB-CONFIDENTIAL			
ease get in touch with the t	emplate owner	EUROPEAN C	ENTRAL BANK					
		EURO	SYSTEM					
EUKOSYSTEM								
Reference date	Legal Entity Identifier	Institution residence country	Collateral Type	Nominal	Fee	Remarks		
Reference date (DD/MM/YYYY)	Legal Entity Identifier	Institution residence country (2-digit ISO code)	Collateral Type (dropdown list)	Nominal (EUR)	Fee (%)	Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		
	Legal Entity Identifier					Remarks		

Figure 65: Sample reporting template

Figure 66 shows the template with data completed for the first three rows. Please use a similar approach to complete the rest of the data.

nter B Z U +		Wrap Text         General            Merge & Center         Image & Sector         Image & Sector		er Impor Writer Imp er Impor Writer Imp Styles		· · · · · · · · · · · · · · · · · · ·
	on residence country					
В	с	D	E	F	G	н
r fill-in instructions ease get in touch with the ter	nplate owner	EUROPEAN C	ENTRAL BANK	I	ECB-CONFIDENTIAL	
Reference date (DD/MM/YYYY)	Legal Entity Identifier	Institution residence country (2-digit ISO code)	Collateral Type (dropdown list)	Nominal (EUR)	Fee (%)	Remarks
	Legal Entity Identifier	(2-digit ISO code)			(%)	Remarks Sample Text
(DD/MM/YYYY)		(2-digit ISO code)	(dropdown list)	(EUR)	(%)	
(DD/MM/YYYY) 26/01/2019	123456789	(2-digit ISO code) 0 55 2 66	(dropdown list) ABC	(EUR) 2500	(%) 10 11	Sample Text

Figure 66: Sample reporting template with completed data

## 3.2.9 Submitting files in CASPER

CASPER provides two different ways of submitting files. You can upload either a single file (zipped or unzipped) for a specific reporting obligation, or a zip file containing one or more files. The second option is very useful when you have multiple reporting obligations and it is only available for standard data collections (which are not event-based).

In addition, you can upload one explanatory document as an attachment to each submission. See Section 3.2.9.5 for more information on attachments.

There are specific rules on file names and extensions that must be followed:

- the file extensions accepted are: xlsx, xlsm, xlsb, csv, txt, xml, xsd, zip, gz, pdf, docx, mat and msg;
- the characters permitted in the file name are: A-Z, a-z, 0-9, +, -, ~, \$, &, #, @, (, ), [, ], {, } and space;
- the file name cannot have more than one full stop symbol (".").

## 3.2.9.1 Single file upload

Figure 67 shows the single file upload process for non-event-based data collections. Follow the steps below to upload the file.

- 1. Go to the submission list screen, select the reporting obligation for which you would like to submit a file and click on "Actions". A small box pops up containing eight options (shown in Table 14).
- 2. To upload a single file, click on the first option: "Upload file".

Available actions	Explanation
Upload file	Upload a single file for this specific reporting obligation.
Download file	Download the file submitted.
View validation results	View the validation results for the specific file submission.
Download template	Download the reporting template you want to complete and submit.
Upload attachment	Available only if the submission is successful.
Update attachment	Available only if an attachment has been uploaded.
Download attachment	Available only if an attachment has been uploaded.
View validation rules	View the business validation rules applicable for this reporting obligation.

#### Table 14: Available reporting obligation actions



Figure 67: Single file submission process

The system shows a file upload pop-up window with information about the reporting obligation. Please make sure that the file you are submitting belongs

to this reporting obligation. The file can be submitted in either uncompressed or compressed form if the zip folder only contains the file itself. Follow the steps to complete the file upload process as shown in Figure 68.

	File Uploa	ad					×	English
nission l	MODULE	REPORTING CYCLE	REFERENCE DATE	REPORTING CODE	ENTITY NAME	ENTITY TYPE		
of Entries fou	MYMODULE3	REPCYC	02/04/2020	ENT1	Entity 1	Group	d	Екро
= CYCL	File*				<b> </b>	Browse File	3 +≡	
τ	Filled in templa	ate.xisx 2				Testing Purpose	T	O DD/M
0 ТСҮС								30/04/20
1 REPC						4	13	10/05/20
о тсус						Upload Close		30/04/20
0 REPCTO	EIN12	MYMODULES	ENURY 2	02/04/2020	0			10/05/20



- 1. Click on the "Browse" button to select the file.
- 2. Once you have selected the file, you will also see the file name.
- 3. Only check this tick box if you would like to upload a file for testing purposes. For more details on the test submission process, see Section 3.2.13
- 4. Click on the "Upload" button to upload the file. The system shows the progress of the upload until it is complete.
- 5. The system displays confirmation that it has received the file and started the validation process. Click on "Link to file vault" to view the file processing status, as shown in Figure 69.

	File Upload	×	Engli
Entries fou	Validation of the file has started. Please check the File Vault screen for the processing status and validation results.  Link to File Vault	d 3 <b>+</b> ≡	Ex REMIT DATE
т 0 ТСУС 1 REPC	File*     Browse File       Filled in template.xisx     Testing Purpose	<b>T</b>	30/04.
O TCYC O REPC	Upload Close	]	30/04. 10/05.

Figure 69: File upload confirmation screen

The system redirects you to the file vault screen, where you can see all the files you have submitted, along with their processing status. For more information on the file vault, see Section 3.2.11.

The workflow is very similar for event-based data collections. At any time, a data submitter can perform an event-driven ad hoc submission for an available event (i.e. a reporting obligation that has not yet reached the end date) by following the procedure explained in Figure 70



Figure 700: Event-driven submission

- 1. Go to the "Submission List" screen and select the available event for which you would like to perform a submission by clicking on the corresponding radio button.
- 2. Once the event is selected, the user will be able to choose the file for submission by clicking on the "New Event" button.

You can also trigger a submission by right clicking on the available event and selecting "New Event".

Following this procedure, the pop-up shown in Figure 68 will appear. The user will be able to select the submission file by clicking the "Browse File" button.

The user will need to enter the event date of the submission. The submitter can also include a comment using the field "Event Comment".

MODULE	REPORTING CYCLE	FORMAT	REPORTING CODE	ENTITY NAME	ENTITY TYPE
EVENT1	TEST	Excel	DEABCDEFGH0123456789	Test Bank 1	Group
Event Date*		Event C	omment		
DD/MM/YY	YY	<b>—</b>			
File*					Browse File

#### Figure 71: "Submit Event" pop-up

The event date of the submission must lie inside the reporting cycle.

## 3.2.9.2 Zip file upload

The second option for uploading files is a zip file bulk upload. This is a convenient option when you would like to upload more than one file at once. The files may relate to several reporting obligations. To upload a zip file, first find the "Zip Upload" button on the submission list screen (Figure 69). By clicking on it, you will see a pop-up box for a zip file upload, as shown in Figure 70.

				EUROPEAN CEI	NTRAL BANK   EU	JROSYSTEM			Enviro
•				Centralised Sul	bmission Platform	(CASPER)			🞴 👻 🛓 Logou
<del>ن</del> ا					Submissions				
18	Submission	.ist							
ission List	Number of Entries for	nd: 2					Download Files	Zip Upload	Export CSV Reset View
ault ation Results	# = cyci	E = REPORTING CODE	1 *= MODULE 2 *=	ENTITY NAME	REFERENCE DATE	= OBLIGATION STATUS	RECEPTION 3 -=	REMITTANCE 4	ATTACHMENT = ACTION
		T	T	r 🗌 T	O DD / MN / YYYY	τ		O DD / MN / YYYY	т
sions									
rts	0 RC1	RE01	MOD1	test	13/04/2021	0		14/04/2021	I

Figure 72: Zip file submission process

Click on the "Browse" button and select the zip file. When you select the file, you will see the file name displayed in the pop-up box, as shown in Figure 70. Check the file name to ensure it follows the naming convention.

Unlike a single file upload, all Excel files in a zip file need to follow a specific file naming convention, otherwise the system will not be able to link the files to the corresponding reporting obligation. The format for the naming convention is:

FileType\_DataCollection\_ReportingCycle\_Module\_ReportingEntity\_Refere nceDate\_SequentialNo\_EntityType

The attributes used in the naming convention are explained in Table 14.

yn l	Zip File Upload		×
es fou	Zip File*	Browse File	d
CYCL	t-casper.escb.eu.zip		3 ₊≡
			<b>T</b>
TCYC REPC		Upload Close	] )7

#### Figure 73: Zip file upload pop-up window

CASPER allows a maximum of 100 MB per file for every upload. If you are using the zip upload functionality, make sure that the total size of the zip file does not exceed 100 MB.

As a data submitter, you have the option of submitting multiple files for a given reporting obligation. This process is called "resubmission". Resubmissions can be distinguished by their sequential number, which needs to be set for files uploaded in a zip file. The first file you submit will have the sequential number 1. All subsequent files for that reporting obligation require a higher number in order to be accepted by CASPER.

Table 15 explains the attributes used in the naming convention.

Attribute	Explanation
File type	General type of the file being submitted. In CASPER release 1.2 the only possible values are: • "C" for an actual submission;
	• "TC" for a test submission.
Data collection	The code for the data collection; found on the left-hand side below the top navigation bar when a collection is selected.
Reporting cycle	The name of the reporting cycle; found in the "cycle" column in the submission list view.
Module	The code for the module; found in the "module" column in the submission list view.
Reporting entity	The abbreviation of the reporting entity, also known as the reporting code; found in the "reporting code" column in the submission list view.
Reference date	The reference date; found in the "reference date" column in the submission list view. Please note that for zip uploads the reference date needs to follow the pattern YYYYMMDD.
Sequential number	When uploading files using a zip file it is important that you provide the correct sequential number in the file name. This must be higher than the sequential number of the last file submitted/resubmitted for that specific reporting obligation. The maximum sequential number value supported by CASPER is 99. The value can be found in the "sequential number" toggle column in the submission list view.

Attribute	Explanation
Entity type (optional)	The type of reporting entity, e.g. company (solo), group (consolidated) or person. The entity type can be found in the "entity type" toggle column in the submission list view. Use abbreviations such as:
	CO (company);
	• GR (group);
	• PE (person);
	as required.

Table 15: Naming convention attributes

Each file in the zip file needs to follow the naming convention format. The zip file itself does not need to follow the naming convention format.

Once you submit the file, you will see confirmation that the validation process for the files has started. Click on "Link to file vault" to check the status of your submission.

## 3.2.9.3 Encrypted and signed files

To ensure the confidentiality and integrity of submitted data, CASPER supports encrypted and/or signed modules. These particular modules only accept encrypted and/or signed submissions and can be easily identified by the

encryption icon ( $\stackrel{\frown}{\bullet}$ ) and the signed icon ( $\stackrel{\clubsuit}{\bullet}$ ) in the submission list screen next to the module name, as shown in Figure 71. For further information and a step-by-step guide on how to encrypt or sign a file, please contact the CASPER Control Team.

Sub	miss	ion List				
Num	nber of Ent	ries found: 38	C			
	# ≡	CYCLE	REPORTING 1 =	MODULE T 2 *	ENTITY NAME =	REFERENCE DATE =
	•		τ	CSVENC	τ	OD / MM / YYYY T
	× 13		123	CSVENC	Central Bank of Cy.	01/09/2022
	~ 6		123	CSVENCSIGN 🔒 🏶	Central Bank of Cy.	01/09/2022

Figure 74: Encrypted and signed modules in submission list view

## 3.2.9.4 Upload limits

CASPER has different file size limits depending on the file reception method and the module setup. The bulk upload described in Section 3.2.9.2 has a maximum file size of 100 MB (this refers to the total size of the zip folder, irrespective of the original size of the files included). The maximum file size for the individual file upload (described in Section 3.2.9.1) depends on the setup that the collection owner has put in place.

If you submit an unzipped file which is rejected because of the size limit, you can zip it to reduce its size and submit it again.

Upload type	Maximum <b>raw/uncompressed</b> file size	Maximum <b>individual</b> <b>file upload</b> size (uncompressed or zipped)	Maximum <b>zip file bulk</b> <b>upload</b> size
Upload via	UI or API	UI or API	UI only
Excel	Up to 100 MB	Up to 100 MB	Up to
CSV Standard	Up to 100 MB	Up to 100 MB	100 MB

CSV Lean	Up to 650 MB	Up to 500 MB	
CSV Extra Lean	Up to 2560 MB	Up to 1024 MB	
External	Up to 1024 MB	Up to 1024 MB	

Table 16: Upload limits

## 3.2.9.5 Adding an attachment to a submission

CASPER makes it possible to upload attachments for each file successfully submitted. This functionality is available only if no attachment has yet been uploaded for the submission concerned. Attachments may be added to a submission to provide the collection owner with further details, or an explanation of the data submitted.

There are specific rules on the file name and extension that must be followed:

- the file extensions accepted are: xlsx, xlsm, xlsb, csv, txt, xml, xsd, zip, gz, pdf, docx, mat and msg;
- the characters permitted in the file name are: A-Z, a-z, 0-9, +, -, ~, \$, &, #, @, (, ), [, ], {, } and space;
- the file name cannot have more than one full stop symbol (".").

Note that only one attachment per submission is supported. Once uploaded, the attachment cannot be modified or removed.

Figure 75 shows how to start the attachment upload process. Follow the steps below to upload the file.

- 1. Go to the submission list screen, select the reporting obligation for which you would like to submit an attachment and click on "Actions". A small box with the available options will appear.
- 2. To upload an attachment, click on "Upload attachment".

										Submission F		EUROSYSTEM m (CASPER)						0	- 14	Environ Logout EN
										Su	brnissior	8								
	Sut	omi	ssic	on List																
omission List	Nun	nber o	f Entri	es found: 2										Download Files		Zip Upload	Бфо	rt CSV	Rese	t View
Vault dation Results			=	CYCLE	=	REPORTING 1 *=	мо	DULE 2 *	-	ENTITY NAME	=	REFERENCE DATE	=	RECEPTION 3 -	=	REMITTANCE	4 .=	ATTACHMEN	NT :	ACTIO
tussions			τ		τ	τ			τ		۲	O DD / MN/ YYYY	۲	· · ·	τ	ODD/ MN/ YYYY	۲			r   (
rts			1	RC1		RE01	м	DD1		test		13/04/2021		11/05/2021 19:55:46		14/04/2021				1
			0	RC1		RE01	M	DD2		test	Downlo Upload		2	2		14/04/2021				I

Figure 75: Attachment upload process

The system shows a file upload pop-up window. This displays the details of the reporting obligation concerned. Please make sure that the file you are submitting relates to this reporting obligation. Follow the steps to complete the file upload process as shown in Figure 76.

Attachm	ent Upload					
MODULE	REPORTING CYCLE	REFERENCE DATE	REPORTING CODE	ENTITY NAME	ENTITY TYPE	SEQ. NO.
MYMODULE3	REPCYC	02/04/2020	ENT1	Entity 1	Group	2
Attachmer	it*				Brov	wse File
Explanatory d	locument.txt 2					
					Uploa	d 3 close
MODULES	Enuty 2	02/04/2020	0	_	_	10/05/2020

Figure 76: Attachment Upload pop-up window

- 1. Click on the "Browse" button to select the file.
- 2. Once you have selected the file, its name is displayed.
- 3. Click on the "Upload" button to upload the file. The system shows the progress of the file upload until it is completed.
- 4. The system displays confirmation that the file has been received and will be displayed in the submission list (Figure 74), from where the collection owner can download it.
- 5. It is possible to update an existing attachment by clicking on "Update attachment" in the "Actions" menu of the submission list.

MODULE     REPORTING CYCLE     REFERENCE DATE     REPORTING CODE     ENTITY NAME     ENTITY TYPE     SEQ. NO.       MYMODULE3     REPCYC     02/04/2020     ENT1     Entity 1     Group     2       Attachment*       Explanatory document bxt	•	The upload of the att	achment was succes	sful and will be displa	ayed in the subm	ission list short	ly.
Attachment* Browse File	MODULE	REPORTING CYCLE	REFERENCE DATE	REPORTING CODE	ENTITY NAME	ENTITY TYPE	SEQ. NO.
	MYMODULE3	REPCYC	02/04/2020	ENT1	Entity 1	Group	2
Explanatory document.txt						Brow	wse File
	Explanatory of	locument.txt					

Figure 77: Attachment upload confirmation screen



You can identify submissions with attachments from the value in the "Attachment" column in the submission list (Table 6) and in the file vault (Table 16).

## 3.2.10 Processing files in CASPER

As a data submitter, you have to submit data files and check that the submission has been successful. You need to understand the CASPER file processing procedure so that you can check whether your submission has been successful. This section explains how CASPER processes files, the individual steps in this process and the corresponding statuses in the file vault, and what they mean.

Once you submit a file to CASPER, the system starts the validation process and checks that the file meets all the requirements for acceptance and processing.

Figure 78 provides an overview of file processing in CASPER, showing the various steps and final statuses.



Figure 78: File processing: steps and final statuses

The table below explains the individual steps in the CASPER file processing procedure.

- 1. When you submit the file to CASPER, the technical validation process is started. Among other things, it checks that the file extension is valid, that the file size is within the set limits and that the file is not corrupt.
- 2. If technical validation fails, the system rejects the file and updates the processing status to "File rejected". For information on how to access the technical validation results, see Section 3.2.12.
- 3. If you cancel the file processing while it is underway, the system updates the processing status to "Cancelled".
- 4. If technical validation is passed, the system starts the business validations. If a fatal error occurs during this process, the system rejects the file and updates the processing status to "Rejected". For information on how to access business validation results returning a fatal error, see Section 3.2.12.
- 5. If the business validation process is completed successfully, the system updates the processing status to "Completed". For information on how to navigate to the submission list page to retrieve validation results, see Section 3.2.12.

## 3.2.11 Check file vault

In the file vault you will find all files you have submitted for your reporting obligations, irrespective of their processing status.

Please note that you may not be able to see files submitted in the past for a reporting cycle that is now closed. The collection designer can hide obsolete data in the file vault.

There is more than one way to navigate to the file vault screen. On the submission list screen, you will see a highlighted panel on the left-hand side. Click on "File vault" to access it. Alternatively, the file vault can be accessed by clicking on "Link to file vault" when processing a submitted file (see Section 3.2.9). Figure 80 below shows the file vault screen.

												ubmission Plat										8	- 1	Envire Logout
5										त्र हो संस्था		Submis	sions											
a de la del	File	Vault																						
ubmission List		ber of Entries found: 4																Do	mento	od Ries	Бф	ort CSV	Ret	set View
ailiauon nesulis		FILENAME	1	MODULE	•	REPORTING	•	ENTITY TYPE	•	ENTITY NAME	•	REFERENCE DATE	-	REMITTANCE DATE	•	RECEPTION DATE 1 + II	ORIG./PAI		-	PROCESSING STATUS	•	VALIDATION STATUS	=	ACTIONS
Necussions			۲		۲		۳		۲		۲	0 00 / MV/ mm	۲	0 00 / MV/ YYYY	۳	T			۳		۳		۳	
ports	0	C_100005_RC1_M	00	MOD1		RE01		Company		test		13/04/2021		14/04/2021	1	11/05/2021 19:55:46	Test Subr	nission xis	sx	Completed	_	Valid	_	1
		C_100005_RC1_M	00	MOD1		RE01		Company		test		13/04/2021	Dow	load Validation Results		04/05/2021 09:24:32	100005_N	10D1_1_2	20	File Rejected		NA		1
		C_100005_RC1_M	00	MOD1		RE01		Company		test		13/04/2021	Cano		Г	04/05/2021 08:31:27	Submissio	nxisx		File Rejected		NA		1
		C_100005_RC1_M	00	MOD1		RE01		Company		test		13/04/2021		14/04/2021		26/04/2021 19:36:54	Submissio	on File xisx	ĸ	File Rejected		NA		1

#### Figure 79: File vault screen

The file vault displays all files submitted by your organisation, even if the file status is "Rejected", the obligation is optional, or the cycle is now closed. In the submission list, however, you will not see entries for file submissions with the status "Rejected" that are no longer being processed, or entries relating to optional reporting for which the cycle is closed. This is the main difference between the two screens.

Table 17 provides an overview of all file vault list attributes in the default view.

Attribute	Explanation
File name	The name of the file using the CASPER naming convention. If you upload a single file via the submission list where no file naming convention is required, the system generates this name automatically.

Attribute	Explanation
Module	The code of the module as established by the collection owner.
Reporting code	The code which identifies the reporting entity.
Entity type	The type of reporting entity, i.e. group, company or person.
Entity name	The name of the reporting entity.
Reference date	The date to which the reported data refer. The field is empty for event-based data collections.
Remittance date	The due date for file submission. This column will not be shown for event-based data collections.
Reception date	The date on which the system received the file.
Orig./parent filename	For a single file upload, this shows the filename as per the naming convention. For a zip file upload, it shows the name of the zip file.
Processing status	The processing status of the file (see Figure 75).
Actions	The list of actions you can perform for every file:
	<ul> <li>download submitted file (see Section 3.2.17);</li> </ul>
	<ul> <li>view validation results for file;</li> </ul>
	cancel submission.

Ta		ault attribut	•		

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table 18.

Attribute	Explanation
Module version	The version number of the module used in the reporting obligation.
Username	CASPER username of the data submitter.
Transmission channel	U2A (user to application) if the file has been manually uploaded via the UI.
File size (KB)	File size in kilobytes.
Validation status	File status after the business validation rules have been processed. Depending on the highest severity level of all the rules failed, the status may be:
	• warning;
	• error;
	• fatal.
	If all rules are passed, the status will be:
	• valid.
	If no business validation rules are executed, the status will be:
	• NA.
Test file	Indicates whether a file has been uploaded using the testing functionality (as explained in Section 3.2.9.1. Possible values:
	• Y (yes);
	• N (no).
Sequential number	The sequential number of the file (as explained in Section 3.2.9.2).
Revalidation date	Timestamp of the revalidation. Blank if the file has not been revalidated.
Processing start time	Timestamp indicating when CASPER launched the processing of the file.

Attribute	Explanation
Processing end time	Timestamp indicating when CASPER ended the processing of the file.
Attachment	Indicates whether an attachment to the submission has been uploaded. Possible values:
	• Y (yes);
	• N (no).
Event Identifier	This is the identifier of the event. This column will not be shown for non-event-based data collections.
Event Comment	This shows any comment for the event. This column will not be shown for non- event-based data collections.

Table 18: File vault list – additional attributes

In the file vault screen, you can see the status of a file submission in the "Processing status" column. The file submission process has intermediate statuses as well, which change throughout the course of the submission process. Table 19 explains these.

File processing status	Explanation
File received	CASPER has saved the file.
File in process	The file has been selected for technical validations.
File rejected	Technical validation failed.
File processed	Technical validation passed.
Pending business validation	Pending execution of business validation.
Performing business validations	Displayed when business validations are running.

File processing status	Explanation
Rejected	Business validation failed as the file failed a validation rule with a severity level of "fatal".
Completed	File successfully processed.
Pending business revalidation	The execution of business revalidation is pending.
Performing business revalidations	Displayed when business revalidations are running.
Pending external validation	Waiting for answers from an external business validation (i.e. not performed by CASPER).
Pending external revalidation	Waiting for answers from an external business revalidation (i.e. not performed by CASPER).
Request cancel	The user has made a request to cancel the action.
To be cancelled	The system has started working on a cancel action request.
Cancelled	The RA has decided to cancel.
Discarded	Deleted by the system.

#### Table 19: File processing statuses

## 3.2.12 Checking technical and business validation results

CASPER enables you to view validation results on the validation results screen so that you can check any errors in your submissions.

The system validates data submitted against the business validation rules for your collection and identifies all failed validation rules, including those relating to the values calculated. This helps you to identify which data points have caused the failure (Figure 78).

Once you are on the submission list screen, select the reporting obligation for which you would like to check the validation results, click on "Actions" and select "View validation results", as shown in Figure 76.

od voge	Subr	nissio	on List																		
lections	Numbe	er of Entri	es found: 124												Download File		Zip Upload		Export CSV	Re	set View
nission List		÷ -	CYCLE	=	REPORTING	1.12	MODULE 2*	=	ENTITY NAME	=	REFERENCE DATE	=	OBLIGATION STATUS	-	RECEPTION 3 -	-	REMITTANCE	4	ATTACHMENT	=	ACTIONS
Vault		۲ 🗌		۲		۲		۳		۲	O DD / MN/ YYYY	۲		۲		۲	0 00 / MN/ YMY	۲		۲	
ation Results		0	ACC01		ENT		ACC01		Entity		01/03/2021		0				02/04/2021				1
ssions TS		0	EXLEAN		ENT		CLOSEDTABLE		Entity		06/04/2021		0				29/04/2021				1
		0	CYCLE2		ENT		CLOSEDTABLE		Entity		20/02/2021		0				05/03/2021				1
		~ 5	EXLEAN		ENT		CLOSEDTABLE		Entity		07/04/2021		A		09/04/2021 18:24:38		01/05/2021				1
		1	5		ENT		DAILY		Entity		05/03/2021		۲		Upload File		25/03/2021				1
		0	EXLEAN		ENT		EXLEAN01		Entity		05/04/2021		0		View Validation Results	;	3 /04/2021				1.1
		0	EXLEAN		ENT		EXLEAN02		Entity		05/04/2021		0		Revalidate Download Template		29/04/2021				1
		ν 2	EXEDGE		ENT		EXLEANEDGE		Entity		11/04/2021		A		Upload Attachment Download Attachment		31/05/2021				1
		1	BULK		ENT		EXLEANTEST		Entity		08/04/2021		<b>A</b>		View Validation Rules		30/05/2021				1

#### Figure 80: Navigating to validation results

You can also access technical and business validation results from the file vault screen.

Depending on the type of validation failed, the system redirects you either to the technical validation results screen or the business validation results screen. Figure 81 shows the technical validation results screen. The error message for the technical validation is highlighted in the yellow box. You can download the full list of technical validation results in CSV format by clicking on the download button in the pop-up window.



#### Figure 81: Technical validation results screen

Figure 80 highlights different areas of the screen. The highlighted box shows the key attributes for the reporting obligation. In the grid list, you can see all the validation rules which have failed using the data submitted. Table 20 gives an overview of all validation result list attributes in the default view.

								CENTRAL BANK								Environ
						Cent	tralised	Submission Platfor	n (CAS	PER)						<b>03 4</b> leg
9					Submissions							U	iser & F	Roles		
eduals.	Validation	n Re	sults for f	Sub	mission											
mission List •	Cycle					Module Code			Repor	rting Entity Code				Seq. No.		
idation Results	TEST					EXT			ENT					1		
raun	Reference Date					Reception Timestamp			Reval	idation Timestar	np			Reporting Entity Na	me	
lation Results	08/04/2021					06/04/2021 15:24:42								ENTITY1		
orts	Number of Entri	es four	vd: 0												Export CSV	Reset View
	RULE ID 1 *	=	ERROR	=	FORMULA		-	FORMULA LEFT VALUE	=	OPERATOR		FORMULA RIGHT VALUE	=	ERROR MESSAGE	2.	= ACTIONS
		۳		۲			۳		T		۳		٣			T

Figure 822: Business validation results screen

The business validation results screen does not display the validation rules that have passed, only those that have failed. By contrast, the validation rules screen (described in Section 3.2.7) displays all validation rules applicable to a submission.

Attribute	Explanation
Rule ID	The unique ID of the validation rule.
Error severity	The severity of an error. There are three levels of error severity:
	• warning;
	• error;
	• fatal.
Formula	The formula for the rule. This is the criterion to be verified when a validation rule is executed.
Formula left value	The actual value on the left-hand side, evaluated based on the formula.
Operator	The comparison operator that compares the content of the left-hand side of the formula with the right-hand side.
Formula right value	The actual value on the right-hand side, evaluated based on the formula.
Error message	The error message if the validation rule fails. The message may also include the specific failed observation value.
Actions	The list of actions you can perform for each file. Only one option is available:
	<ul> <li>view the definition of the failed validation rule.</li> </ul>

Table 20: Validation results screen attributes

In addition, you can add toggle columns for the attributes shown in Table 21.

Attribute	Explanation
Difference	The difference between the right-hand and left-hand side values.
Difference %	Indicates whether the difference is a percentage or an absolute value. Possible values: • Y (yes); • N (no).
Interval	The range for value deviation. This is a positive integer.
Interval as %	Indicates whether the interval is a percentage or an absolute value. Possible values: • Y (yes); • N (no).
External rule	Indicates whether the failed rule is external, i.e. not performed by CASPER. Possible values: • Y (yes); • N (no).

Table 21: Validation results screen attributes

You can view the definition of every failed validation rule by clicking on "Actions" and then "View". This opens a read-only pop-up window and shows key attributes of the validation rule, the evaluation results with the actual values of the left-hand and right-hand side of the formula as well as a separate section with the actual formula. Figure 83 shows the pop-up window.

Rule (P*) Severity   R04 WMRNING External Rule   Formula Lett Value 0 Efference   3 4 3 Difference as %     Formula Lett Side* Formula Operator*   Formula Lett Side* Formula Operator*   Formula Right Side*   DPOLNUMESEQ_OPT Image: Comparison of the second rule	A Basic Details				
3 4 1 Difference as %  Formula Left: Gide* DP(NUMBER2_OPT)  Error Message Definition* Data point is less than 4. Please check NUMBER2_OPT Interval			-	e	
Formula Left Side* Formula Operator* Formula Right Side* DPINNUMBER2_OPT  Form Message Definition* Data point is less than 4. Please check NUMBER2_OPT Interval					
Formula Idet Side*         Formula Operator*         Formula Right Side*           DP00UNDEEX2_OPT)         >=         4           Error Message Definition*             Data point is less than 4. Please check NUMBER2_OPT             Interval	3	4	1		Difference as %
Formula IdeK Side* Formula Operator* Formula Right Side* DPRNUMBER2_OPT)  Form Message Definition* Data point is less than 4. Please check NUMBER2_OPT Interval Interval	▲ Formula				
DPINUMEER2_OPT) 4 Error Message Definition* Data point to less than 4. Please check NUMEER2_OPT Interval		Formula Opera	ator*	Formu	ula Right Side*
Data point is less than 4. Please check NUMBER2_OPT	UPILINUMBERC_UP1)	24		4	
Interval					
	Data point is less than 4	I. Please check NUMBER2_OP	T		
0 Interval As %					
	0	Interval As %			

Figure 83: Validation results pop-up window

## 3.2.13 Discussing failed validations

CASPER runs business validation rules to verify the data quality of a submission. For each failed validation, an error message is displayed along with the corresponding validation rule.

If a failed validation requires clarification or additional documents to be resolved, you can use the discussion functionality. This enables data submitters and collection owners to discuss failed business validations and exchange related documentation so they can settle the issue and fulfil the reporting obligation.

	Cer	tralised Submission Platform (CASPER)		😪   🔤   🛎 Logout 📑
S.	Submissions	1	User & Roles	
Data Collections				
Number of Entries found: 1			I	Export CSV Reset View
CODE T	NAME	DESCRIPTION		ACTIONS
4				- 2
hand yough	DC100005	Test		<b>6</b>
				View Submission List View File Vault
				View Validation Results
				View Discussions
				Manage Email Subscription

Figure 84: Navigating to the discussions screen

- 1. Click on the Submissions menu at the top of the secondary navigation pane.
- 2. The submissions screen displays the list of data collections assigned to you. Select a data collection and click on its "Actions" button. A small popup box with five options appears:
  - View submission list;
  - View file vault (see Section 3.2.11);
  - View validation results (see Section 3.2.12);
  - View discussions (see Section 3.2.13);
  - View reports (see Section 3.2.14).
- 3. Click on the "View discussions" option to go to the discussions screen shown in Figure 85.

Construction of the local division of the lo	Discussi	UIIS																
Submission List	Number of Er	cries four	d. 95														Export CSV	Reset View
File Vault Validation Results	CYCLE		MODULE 2 *	•	REPORTING	1 *#	ENTITY NAME		REFERENCE DATE	3 +11	REMITTANCE DATE	4	OBLIGATION STATUS	-	PENDING ECB INPUT		PENDING RA INPUT	
Discussions		۳		۳		۳		۲	0 00 / MN/ YYYY	۲	0 00 / MV/ YYYY	۲		۲		۲		
Reports	TEST		EXTRALEAN		ENTITY1		ENT/TY1		16/04/2021		16/04/2021		<b>A</b>		2			
	TEST		EXTRALEAN		ENTITY1		ENTITY1		15/04/2021		21/04/2021		<b>A</b>					
	SANITY		INVALID		ENTITY1		ENTITY1		22/04/2021		29/04/2021		•					

Figure 855: Discussions screen

The discussions screen will show all your reporting obligations for which at least one validation rule has failed with a severity level of "error" or "warning".

For resubmissions, only the most recent submission (with the highest sequential number) is displayed on the discussions screen.

Table 22 provides an overview of the attributes displayed in the default view.

Attribute	Explanation
Cycle	This is the name of the reporting cycle as explained in Table 6.
Module	This is the code of the module, as defined by the collection owner.
Reporting code	This is the code which identifies the reporting entity.
Entity name	This is the name of the reporting entity.
Reference date	This is the date to which the reported data refer. It will display the event date for event-based data collections.
Remittance date	This is the due date for file submission. This column will not be shown for event- based data collections.
Obligation status	This is the obligation status (Table 8).
Pending ECB input	Indicates that the collection owner's input on one or more discussion topics is pending. A data submitter has added the latest comment and the discussion does not have "accepted" status. Possible values:
	🐱 input pending;
	no input required.

Attribute	Explanation
Pending RA input	Indicates that your input on one or more discussion topics is required. The collection owner has added the latest comment and the discussion does not have "accepted" status. Possible values:
	🔽 input pending;
	no input required.
Accepted	Indicates whether the discussion has been concluded by the collection owner. Possible values:
	<ul> <li>Y (yes) – the collection owner has accepted the data submitter's comments;</li> </ul>
	<ul> <li>N (no) – the discussion is ongoing.</li> </ul>

Table 22: Discussion screen default columns

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table 23.

Attribute	Explanation
Entity type	The type of the reporting entity, i.e. company, group or person.
Entity group	The name of the reporting entity group, as established by the collection owner.
Number of errors	Number of failed validation rules with a severity level of "error" that have occurred during the business validation process.
Number of warnings	Number of failed validation rules with a severity level of "warning" that have occurred during the business validation process.

Attribute	Explanation
Event Identifier	This is the identifier of the event. This column will not be shown for non-event-based data collections.
Event Comment	This shows any comment for the event. This column will not be shown for non- event-based data collections.

#### Table 23: Discussion screen additional columns

For each validation error, you can check the validation result for the data submitted and the validation rule to determine the root cause (see Section 3.2.12). If you have checked everything and conclude that the data submitted are correct and you would therefore like to receive clarification about the result, you can begin a discussion with the collection owner by following the steps below. Note that the collection owner may also initiate a discussion.

111	Discussio	ns															
ubmission List	Number of Ent	ies four	nd: 95														Export CSV Reset Vie
ile Vault alidation Results	CYCLE		MODULE 2*		REPORTING 1 *		ENTITY NAME	REFERE DATE	4CE	3 •0	REMITTANCE	4.0	OBLIGATION STATUS		PENDING ECB INPUT	- 1	PENDING RA INPUT
Necussions		۲		۲	· · · · ·	۲ [	т	000/	ev/ mo	τ.	0.00/MV/1111	•		۲		۲ (	
leports	TEST		EXTRALEAN		COMP4	8	ENTITY1	16/04/	021		16/04/2021		4				
	TEST		EXTRALEAN		scuss ew Submission List	8	ENTITY1	15/04/	021		21/04/2021		<b>A</b>				
	SANITY		INVALID				ENTITY1	22/04/	021		29/04/2021		▲				

Figure 86: Navigating to the discussion board screen

- 1. On the discussions screen, select the reporting obligation containing the error you would like to discuss and click on its "Actions" icon. A small popup box with two options will appear:
  - discuss;
  - view submission list.
- 2. Click on the "Discuss" option to navigate to the discussion board screen shown in Figure 88.

	Discussio	n B	pard									
ubmission List	Reportin	g Oblig										
le Vault	Entity Name		Reporting Code		Entity Group	Status				Cycle		
alidation Results	ENTITY1		ENTITYI		ESCB1	Received - Error				TEST		
Nscussions •	Module Code		Reference Date	A	Acceptance Comment*					Accepted		
Discussion Board	EXTRALEAN		16/04/2021							No		
	RULE ID	-	ENON MESSAGE	- 1 T	FORMULA		-	ERROR SEVERITY	-	TOPIC STATUS	ļ	ACTIONS
	EXL4		Column 16 must contain an integer		ROUND(DPS(c2d_NumberOfAggregatedDebtors)) = DP	S(c2d_NumberOfAggregatedDebtors)		WARNING		•		1
	EXL98		The values reported in column 16 cannot be all set to the same value. P	P1 1	MIN(DPS(c2d_NumberOfAggregatedDebtors)) < MAX(D	PS(c2d_NumberOfAggregatedDebtors))		WARNING				1
	EXL60		Column S must have value "pool"		DPS(c2d_TypeOfSystem0) = "pool"			ERROR				1
					DPS(c2d_DateOfMaturity) > d01/01/2020			WARNING		•		
	EXL52		column 9 ≻d01/01/2020		or acco_baceonination (y) > do no monzooo					-		

Figure 87: Reporting obligation panel and discussion topics panel

Initially, the discussion board screen consists of two expandable panels. The reporting obligation panel at the top contains the obligation details and information about the overall acceptance of the discussion.

Attribute	Explanation
Entity name	The name of the reporting entity.
Reporting code	The code which identifies the reporting entity.
Entity group	The name of the reporting entity group, as established by the collection owner.
Status	The obligation status (Table 8).
Cycle	The name of the reporting cycle, as explained in <b>Error! Reference source not found.</b>
Module code	The code of the module, as established by the collection owner.
Reference date	The date to which the reported data refer.
Acceptance comment	The collection owner's comment on acceptance of the discussion (maximum length: 250 characters). Mandatory if the "accepted" field is set to "Yes".

Attribute	Explanation
Accepted	Indicates whether the discussion has been concluded by the collection owner. Possible values:
	<ul> <li>Y (yes) – the collection owner has accepted the data submitter's comments;</li> </ul>
	<ul> <li>N (no) – the discussion is ongoing.</li> </ul>

Table 24: Reporting obligation panel attributes

The discussion topics panel shows the validation rules for the obligation that has failed and their status ("Topic Status") (see Table 25).

Attribute	Explanation
Rule ID	The unique ID of the validation rule.
Error message	The error message, as established by the collection owner.
Formula	The formula for the rule. This is the criterion that is verified when a validation rule is executed.
Error severity	The severity of the error. As fatal errors are not displayed, there are two possible values:
	• warning;
	• error.

Attribute	Explanation
Topic status	The status of the individual validation rule discussion. Possible values:
	"new" (initial status when a validation has failed);
	"ongoing" (when a comment has been added OR the same validation failed in a previous submission AND the collection owner has not yet changed the status to "accepted");
	<ul> <li>"accepted" (when the collection owner has changed the status to "accepted");</li> </ul>
	"resolved" (when failed validation results have been resolved in a subsequent resubmission).
	Note that the default filter settings for this column exclude discussion topics with "resolved" status. Change the filter to view resolved topics as well.
Actions	The list of actions you can perform for each validation result. Two options are available:
	<ul> <li>view comments (opens the comments panel);</li> </ul>
	<ul> <li>view validation results (navigates to the validation results screen; Section 3.2.12).</li> </ul>
Table 25: Discussion topics panel columns	

Table 25: Discussion topics panel columns

Note that a new discussion topic will be created automatically for each failed validation in a reporting obligation. If the same validation fails in a resubmission, no new discussion is created.

No.	RULE ID		FORMULA	ERROR SEVERITY	TOP	HC STATUS		ACTIONS
	EQ.4	Column 16 must contain an integer	ROUND(DPS(c2d_NumberOfAggregatedDebtors)) = DPS(c2d_NumberOfAggregatedDebtors)	WARNING				1
ibmission List e Vault	EXL98	The values reported in column 16 cannot be all set to the same value. PL.	MIN(DPS(c2d_NumberOfAggregatedDebtors)) < MAX(DPS(c2d_NumberOfAggregatedDebtors))	WARNING		9		1
Idation Results	EX1.60	Column 5 must have value "pool"	DPS(c2d_TypeOfSystem0) = "pool"	ERROR		9		1
cussions •	EXL52	column 9 >d01/01/2020	DPS(c2d_DateOfMaturity) > d01/01/2020	WARNING		9		1.1
liscussion Board	EXL50	column 9 >d01/01/2010	DPS(c2d_DateOfMaturity) > d01/01/2010	WARNING		•		1 L
	Comments     Number of Entries fo     DATE	und: 0 USER E COMMENT				Export CSV ATTACHMENT	_	t View
		T T T			T		T	
			No search result found					
	Accepted	Commerce* This is a cest commerc.						Cancel

#### Figure 88: Comments panel

If you select a rule or click on its "Actions" icon and select "View comments", the comments panel (a discussion thread with the collection owner) will appear below. Discussion participants can also attach explanatory documents to their comments. Table 26 provides an overview of attributes displayed in the default view.

Attribute	Explanation
Date	The timestamp for when the comment was added.
User	The user ID of the user who added the comment.
Comment	The comment, with a maximum length of 500 characters.
Attachment	Indicates whether an attachment has been added to the comment. Possible values: (Yes); (No).

Attribute	Explanation
Actions	The list of actions you can perform for each validation result. three options are available:
	<ul> <li>add attachment (enabled for your comments only);</li> </ul>
	<ul> <li>update attachment (enabled only if there an attachment has already been uploaded);</li> </ul>
	<ul> <li>download attachment (if provided).</li> </ul>

#### Table 26: Comments panel default columns

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table 27.

Attribute	Explanation
Username	The name of the user who added the comment

#### Table 27: Comments panel: additional columns



Figure 89: Add a comment to a discussion

### To add comments to a discussion, follow the steps in the table below.

- 1. On the discussion board screen, select the validation rule you would like to discuss. The comments panel will appear below.
- 2. Add your comment in the Comment field. The "Accepted" checkbox is deactivated as only collection owners can accept a discussion.
- 3. Save the comment by clicking on the "Save" button in the lower right-hand corner of the screen. A success message is displayed at the top of the screen. The topic status changes from "new" to "ongoing".

III         ERROR         III           IIII         IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
PSTerm_Typel), "TERM"], NOT ISNULL (DPSGestement_Date ERROR P I StColumeral_Typel), "SECURITES', "CASH" + TRUE ERROR P I Inn_Codel) + 13 ERROR P I PSTerm_Typel), "DRN", NOT ISNULL (DPSG/anurity_Datel) ERROR P I IPSTerm_Typel), "OPEN", NOT ISNULL (DPSG/anurity_Datel) ERROR I I DPSTerm_Typel), "OPEN", NOT ISNULL (DPSG/anurity_Datel) ERROR I I
KCollateral_Typell."SECURTIES", "CASH") = TRUE         ERROR         I           ain_Codell <13
ain_Codell +13 ERROR P I IPS/Term_Typell, "TERM"I, NOT ISNULL (DPS/Maruring_Datel) ERROR P I IPS/Term_Typell, "OPEN"I, (NOT ISNULL (DPS/Reference_Date ERROR I I ISTORMENT, Typell, "OPEN"I, (NOT ISNULL (DPS/Reference_Date ERROR I I EXPORT CSV
PSTerm_Typell, "TERM"), NOT ISNULL (DPSR/eference_Date ERROR 🗢 i IPSTerm_Typell, "OPEN"), (NOT ISNULL (DPSR/eference_Date ERROR 🍬 I Export CSV
PSTerm_Typell, "OPEN"1 (NOT ISNULL (DPSReference_Date ERROR I
Digent CSV
E ATTACHMENT E ACTIONS
т т
ent.
Add Attachment
Providence of the second

Figure 90: Discussion topics panel – add attachment

RULE ID	DATE	USER	COMMENT
R04	03/04/2020 10:40:52	FREXTDATASBMTRTST1	This is my comm
Attachm	ient*		Browse File
Explanato	ry document.txt		

#### Figure 91: Attachment upload screen

To add an attachment to a comment in a discussion, follow the steps shown below.

- 1. To add an attachment to your comment, click on its "Actions" icon. A small pop-up box with three options will appear:
  - add attachment;
  - update attachment (deactivated if no attachment has been uploaded);
  - download attachment (deactivated if no attachment has been uploaded).

Select "Add attachment" to open a pop-window.

2. Click on the "Browse file" button and select the file you would like to attach. Note that attachments must not exceed a file size of 25 MB. Confirm by clicking the "Upload" button. A success message is displayed in the popup window. The attachment column in the comments panel displays the paper clip icon. There are specific rules on the file name and extension that must be followed:

- the accepted file extensions are: xlsx, xlsm, xlsb, csv, txt, xml, xsd, zip, gz, pdf, docx, mat and msg;
- the characters permitted in the file name are: A-Z, a-z, 0-9, +, -, ~, \$, &, #, @, (, ), [, ], {, } and space;
- the file name cannot have more than one full stop symbol (".").

When a discussion has been concluded because, for example, all the validation errors have been resolved, the collection owner may set its status to "accepted" to indicate that no further changes and/or resubmission are required.

Note that there are two levels of acceptance in the discussion functionality. A collection owner can accept discussions at the obligation level (reporting obligation panel) and, individually, at the level of each validation rule (discussion topics panel). Acceptance of all the validation results as a condition for acceptance at the obligation level may vary according to the data collection in question.

#### 

#### Figure 92: Discussions screen navigation

- 1. Click on the "Submissions" menu at the top of the secondary navigation pane.
- 2. The submissions screen displays a list of the data collections assigned to you. Select one and click on its "Actions" button. A small pop-up box with four options appears:
  - view
  - view file vault (see Section 3.2.11);
  - view validation results (see Section 3.2.12);
  - view discussions.
- 3. Click on the "View discussions" option to navigate to the discussions screen shown in Figure 92.

## 3.2.14 Discussing other topics

In addition to discussing failed validations in CASPER, users can reply to discussions that are related to other topics.

								EUROPEAN CEN	TRAL BANK   EURC	SYSTEM								-	Environme
								Centralised Subr	nission Platform (CA	SPER)								<b>\$</b>	🖾   🕹 Logout 🚦
2				Submi	ssions	l.								User & R	oles				
	Discussi	ons																	
mation	Number of En	ntries for	und: 5															Export CSV	Reset View
mission List Vault	CYCLE	=	MODULE 2 -	REPORTING	1 -=	ENTITY NAME	=	REFERENCE 3.=	REMITTANCE 4.8	OBLIGATIO STATUS	N =	PENDING ECB	=	PENDING RA	-	ACCEPTED	•	ACTIONS	
lation Results		۲	۲		۲		۳	O DD / MM / YYYY T	O DD / MM / YYYY T		۲		۳		۲		۲		
ussions	NEWV		ATMOD	CENTITY		centity	_	15/10/2021	15/10/2021	4						N		T.	
rts	ATULS		ATULMOD	CENTITY		centity		26/10/2021	26/10/2021	A						N		1	
	NEWV		ATULMOD	CENTITY		centity		15/10/2021	15/10/2021	4						N		1	
	GR0322		ENRICO	CENTITY		centity		22/03/2022	23/03/2022	4						N		1	
	REG132		ENRICO	CENTITY		centity		17/01/2022	17/01/2022	4		-				N		1	
right 2022, European	Central Bank																Resp	onsible Disclosur	e   Privacy Statem

#### Figure 93: Discussions screen

On the discussions screen, you will find all reporting obligations with at least one failed validation ("error" or "warning" severity level) or other discussion topics opened.

For resubmissions, only the latest submission (highest sequential number) is displayed on the discussions screen.

Table 28 provides an overview of the attributes displayed in the default view.

Attribute	Explanation
Cycle	The name of the reporting cycle as explained in <b>Error! Reference source</b> not found.
Module	The code of the module as explained in Table 6.
Reporting code	The code which identifies the reporting entity.
Entity name	The name of the reporting entity.

Attribute	Explanation
Reference date	The date to which the reported data refer.
Remittance date	The due date for file submission.
Obligation status	The obligation status (Table 8).
Pending ECB input	Indicates that your input on one or more discussion topics is pending. A data submitter added the latest comment and the discussion does not have "accepted" status. Possible values:
	input pending;
	no input required.
Pending RA input	Indicates that a data submitter's input on one or more discussion topics is required. You added the latest comment and the discussion does not have "accepted" status. Possible values:
	input pending;
	no input required.
Accepted	Indicates whether you have concluded the discussion. Possible values:
	<ul> <li>Y (yes) – you have accepted the data submitter's comments;</li> </ul>
	<ul> <li>N (no) – the discussion is ongoing.</li> </ul>
Actions	Discuss; view submission list.

#### Table 28: Discussion screen – default columns

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table

Attribute	Explanation
Entity type	The type of reporting entity, i.e. company, group or person.
Entity group	The name of the reporting entity group as explained in Table 7

 Table 29: Discussion screen – additional columns

Only collection owners can add new discussion topics in CASPER. To add comments to an existing discussion, proceed as follows:

_									ROPEAN CEN										c	2 - 4	Environme
								Ce	entralised Sub	missio	in Platform (CA	SPER)	)							S * 8	Logout
				\$	bmissions											User	& Roles				
	Discussio	ns																			
bmission List	Number of Entr	tes found	1: 6																Export CSV	Rese	et View
e Vault lidation Results	CYCLE T	= мо	DDULE 2*	= REP COD	E	*= ENT	TITY NAME	=	REFERENCE	3 📲	REMITTANCE	4 -=	OBLIGATION STATUS	-	PENDING ECB	=	PENDING RA INPUT	-	ACCEPTED	= "	CTIONS
scussions	cycle1	т		T		۲		۲	O DD / MW/ YYYY	۲	O DD / Mb/ 1111	۲		т		۲		۲		τ.	
ports	CVCLE1	МС	DD1	AT		Ent	sity1		04/05/2020		08/05/2020		٥				2		N		1
	CYCLE1	MO	001	AT		Ent	tity1		27/04/2020		30/04/2020		4						N		1
	CVCLE1	МС	DD1	88		Ent	tity 2		04/05/2020		08/05/2020		٢						Discuss		
	CYCLE1	мо	DD1	BE		Ent	tity 2		27/04/2020		30/04/2020		4					2	View Subr	nission Lis	52
	CYCLE1	мс	DD1	E3		Ent	tity 3		04/05/2020		08/05/2020		4						N		1
													4								

Figure 94: Discussion board screen navigation

- 1. On the discussions screen, select the reporting obligation for which you would like to discuss other topics and click on its "Actions" icon. A small pop-up box with two options appears:
  - discuss;
  - view submissions list.
- 2. Click on the "Discuss" option to navigate to the discussion board screen as shown in Figure 95 below.

•			EUROPEAN CENTRAL BANK   EUROSYSTEM Centralised Submission Platform (CASPER)						🤹 i 🔤	Environm
<i>2</i>		Submissions			t	User & Rol	8			
	Discussion Board									
mation	Reporting Obligation Entity Name	Reporting Code	Entity Group	Status				Cycle		1
Vault		hepoting code	Linky Group	Revalidate	ed - Error			GR0322		-
lation Results ussions • cussion Board	Module Code	Reference Date 22/03/2022								
ints	Discuss Falled Rules Discuss Ot	her Topics								
	Number of Entries found: 1						Add Topic	Export CSV	Rese	t View
	TOPIC KEY 1* =	SUBJECT			START DATE 2*	= то	PIC STATUS 3 *		-	ACTIONS
	т			т		T			۲	
	Topic 1	This is the topic 1			19/04/2022 12:57:26					<b>Z</b>

Figure 95: Reporting obligation panel and Discussion Topics panel

The "Discuss other topics" screen will open, and you will see the status of previous discussions. If there is a discussion open due to failed validations related to this reporting obligation, it will take you directly to the "Discuss failed rules" screen. All you need to do is click on the "Discuss other topics" button.

Attribute	Explanation					
Topic key	The unique ID that serves as an identifier of the discussion topic.					
Subject	Subject of the discussion topic.					
Start date	Date on which the discussion topic was raised.					
Topic status	The status of the specific discussion. Possible values:					
	"ongoing": at least one comment has been added first by the collection owner;					
	"closed": at least one comment has been added first by collection owner and the status of the topic is "closed".					

Attribute	Explanation
Actions	The list of actions you can perform for each validation result. Three options are available;
	<ul> <li>view comments (opens the comments panel);</li> </ul>
	edit topic;
	add comment.

Table 30: Discussion topics panel columns

		_				EAN CENTRAL BANK   EUROSYSTEM					-	Environ	
					Centra	alised Submission Platform (CASPER)						• A Legour	
			Sub	missions				User & Roles					
<u> </u>	-				_			L				1	
1	1 Please enter a positive integer value				DPK	(ABC111) > 0		WARNING		•		.1	
2	This va	lue is larger t	than 2,50	0. Please double-check	DPI(	(ABC112) < DPI(ABC122)		•		1			
12	the diff	erence shoul	id be pos	litive	DAT	<pre>FEDIFF(DPI(Datei2), DPI(Datej2)) &gt; 0</pre>		ERROR				1	
26	failed				DAT	FEDIFF(DPI(DateI9), DPI(DateJ9)) = 0		ERROR		9		- 1	
4													
<ul> <li>Commer</li> </ul>	ts												
Number of Em	ries found: 0									Export CSV	F	eset View	
DATE *	-	USER	=	COMMENT					=	ATTACHMENT	=	ACTIONS	
	т		т						T		т		
	No search result found												
Accepted Commert*													
	1 2 26 6 A Commen Numer of tim DATE *	2 The vell 12 Dre eff 28 failed Comments Numer of Strong Found - 0 DATE * = *	Pressenter a pool     This value is larger     This value is larger     To value is larger     Software in the interval of the interval o	Plase enter a positive integ     This value is larger than 3.05     for the difference should be por     formers     form	Plase entre a positive integravalue     This value is larger than 1.500. Plase adoute check     term     term of themes should be positive     failer     Comments     term     te	Centro Roboticities Peace entre a postive integer value OP 1 Prese entre a postive integer value OP 2 This value is larger mail 260. Prese double-theor. OP 12 the difference should be postive OP 28 faired OP 29 Comments Numer of forms tours 0 Date * USE Converse USE Converse					Centralised Submission Platform (CASPEB)           Solutions         Der & Non           1         Plassentine gesonie integer nalue         DPIAdC113) + 0         V MARING         I           2         This value is larger na 2500 Presis double-thes.         DPIAdC113) + 0         V MARING         I           1         This value is larger na 2500 Presis double-thes.         DPIAdC112) + 0         EREOR         I           2         This value is larger na 2500 Presis double-thes.         DPIAdC112) + 0         EREOR         I           3         faired         DATDINTD/PDIAre(D), DPIAdD12) + 0         EREOR         I         I           3         faired         DATDINTD/PDIAre(D), DPIAdD12) + 0         EREOR         I         I           4         Comments         I <t< td=""><td>Centralised Submission Platform (CASPER)       Centralised Submission Platform (CASPER)       Centralised Submission Platform (CASPER)       Centralised Submission Platform (CASPER)       Centralised Submission Platform (CASPER)       Presenter a plassion reger value      DP(AdCI11) + 0      Centralised Submission Platform (CASPER)       The value is imper thm 2.50      The value is imper thm 2.50      Centralised Submission Platform (CASPER)       Note of flows flows 0       Note of flows flows 0       No seach result flowd       No seach result flowd       Contralised       Contralised Submission Platform       No seach result flowd       Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission         Centralised Submission         Centralised Submission         Centralised Submission</td></t<>	Centralised Submission Platform (CASPER)       Presenter a plassion reger value      DP(AdCI11) + 0      Centralised Submission Platform (CASPER)       The value is imper thm 2.50      The value is imper thm 2.50      Centralised Submission Platform (CASPER)       Note of flows flows 0       Note of flows flows 0       No seach result flowd       No seach result flowd       Contralised       Contralised Submission Platform       No seach result flowd       Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission        Centralised Submission         Centralised Submission         Centralised Submission         Centralised Submission	

Figure 96: Comments panel

If you select a topic or click on its "Actions" icon and select "View comments", the comments panel (3) appears below and shows the ongoing discussion thread between data submitter and collection owner. Discussion participants can also attach explanatory documents to their comments. Table provides an overview of the attributes displayed in the default view.

Attribute	Explanation
Date	The timestamp when the comment was added.
User	The user ID of the user who added the comment.
Comment	The comment (maximum length: 500 characters).
Attachment	Indicates whether an attachment has been added to the comment. Possible values: (Yes)
	(No)
Actions	The list of actions you can perform for each validation result. Three options are available:
	<ul> <li>add attachment (enabled for your comments only);</li> </ul>
	<ul> <li>update attachment (enabled only if an attachment has already been uploaded);</li> </ul>
	<ul> <li>download attachment (if provided).</li> </ul>

Table 31: Comments panel – default columns

In addition to the columns that are visible by default, you can add toggle columns for the attributes shown in Table 32.

Attribute	Explanation
Username	The name of the user who added the comment

Table 32: Comments panel: additional columns

			Centralised Submission Platform (CASPER)			🖓 + 🛔 Logo						
2		Submissions		User & Roles								
		1	1.1.1	(*)	1							
- 1	1	Please enter a positive integer value	DPI(ABC111) > 0	WARNING	•	1						
	2	This value is larger than 2,500. Please double-check	DPI(ABC112) < DPI(ABC122)	WARNING	9	1						
sion List	12	the difference should be positive	DATEDIFF(DPI(Date12), DPI(Date12)) > 0	ERROR		1						
ilt on Results	26	failed	DATEDIFF(DPI(DateI9), DPI(DateJ9)) = 0	ERROR	9	1						
ions •	<					+						
ssion Board	<ul> <li>Comments</li> </ul>											
	Number of Entrie	es found: 0			Export CSV	Reset View						
	DATE *	USER COMMENT			ATTACHMENT	ACTIONS						
		T T			τ	T						
	No watch result found											
	Accepted	Comment* This is my comment. Please see my attachment.			T							
						3						



To add comments to a discussion, follow the steps shown in the table below.

- 1. On the discussion board screen, select the discussion topic you would like to discuss. The comments panel will appear below.
- 2. Add your comment in the Comment field. Check the "Closed" box if you would like to close the discussion topic about the validation result.
- 3. Save the comment by pressing the Save button in the lower right-hand corner of the screen. A success message is displayed at the top of the screen.

							Centralised Submission Platform (CASPER)				9	• 🔺 u	Envir
				Sul	bmissions			User & Roles					ſ
1.												•	
27	fai	led					DATEDIFF(DPi(Date(9), DPi(Date(9))) > 0	ERROR		9		1	
26	26 failed						DATEDIFF(DPI(Datel9), DPI(Datel9)) = 0	ERROR			1		
1 List	12 the difference should be positive				ositive		DATEDIFF(DPI(Datel2), DPI(Datel2)) > 0	ERROR		9			
Results 1	Ph	ease e	enter a positi	ve inte	ger value		DPI(ABC111) > 0	WARNING				1	
s • •	omments											-	•
n Board	er of Entries four	nd: 1								Export CSV	R	eset View	v
DATE	•	-	USER	=	COMMENT				=	ATTACHMENT		ACTIONS	s
		۲		т					т		т		
13/04	/2021 12:42:34		EUCASPER	UAT4	This is my comment.	Please see m	y attachment.					1	1
4										Add Attac Download		-	
Acc	epted 0	Comm	ient*										

Figure 98: Discussion topics panel – add attachment

RULE ID	DATE	USER	COMMENT
3 TradeDate	09/10/2019 15:06:29	10 million (2010)	This is a comme
Attachment	*		Browse File
Explanatory do	cument.pdf		

Figure 99: Attachment upload screen

To add an attachment to a comment in a discussion, follow the steps in the table below.

- 1. To add an attachment to your comment, click on its "Actions" icon. A small pop-up box with three options will appear:
  - add attachment;
  - update attachment (deactivated if no attachment has been uploaded);
  - download attachment (deactivated if no attachment has been uploaded).

Select "Add attachment" to open a pop-window.

2. Click on the Browse File button and select the file you would like to attach. Note that attachments must not exceed 25 MB. Confirm by clicking on the "Upload" button. A success message is displayed in the pop-up window. The attachment column in the comments panel shows the paper clip icon. The reports are prepared by the collection owners. Their main purpose is to provide access to tableau reports, data quality reports, or any other reports accessible via a link for the data collection.



#### Figure 101: Report preview sample

For each report, the following elements are displayed (see Figure 100).

3.2.15 Check reports

On the reports screen in CASPER you will see a set of tiles, each containing text and links to external resources, as shown in Figure 100.

<b>A</b>		NTRAL BANK   EUROSYSTEM bmission Platform (CASPER)	Environment: A
۲	Submissions	User & Roles	
Submission List File Vault Validation Results Discussions Reports	TEST ANNOUNCEMENT This is a fast report. Please find here the test report here Click here		

Figure 100: Reports screen

- 1. The title of the report. If the length of the title is greater than the width of the report displayed, hover your mouse cursor over the title to see the full text inside a tooltip.
- 2. The description of the report. Only the first 150 characters are shown the formatting of the report is not visible.
- 4. A "Click here" button, to view all the details of the report.

If you click on the "Click here" button in one of the reports, a pop-up window will appear, as shown in Figure 102. The window displays details of the report, i.e. the title and the full description. The links for the report are clickable and open in a new tab.

Description:		
Click here to find the first report.		



## 3.2.16 Test submission

CASPER allows you to make a test submission. The system treats the file as test data, completes the file processing and shows you the results of your submission. This is an ideal way to check everything is correct before you submit a file.

It is very simple to upload a test file. You follow the normal file upload process for a single file upload, the only difference being that you need to make sure you check the "Testing purpose" box, as shown in Figure 103.

MODULE	REPORTING CYCLE	REFERENCE DATE	REPORTING CODE	ENTITY NAME	ENTITY TYPE
MYMODULE	TCYCLE	26/03/2020	ENT1	Entity 1	Group
File*					Browse File
Filled in myce	oll template.xlsx				Testing Purpose

Figure 103: Testing submission pop-up

The system shows confirmation that the file has been received and the validation process has started. Go to the file vault screen to check the entry in the list.

Test submissions are only visible in the file vault. You cannot see them on the submission list screen. CASPER only stores test files for 24 hours and then deletes them.

Figure 104 shows an entry for a test submission in the file vault. Since all files are stored using the proper naming convention, note that the name of the test file starts with the prefix TC instead of C.

Submissions User	& Roles																				Em	vironment
MYCOLL My collection	File	e Vau	ult																			
	Nu	mber of	Entries found:	8														voad Ries	Expor	t CSV R	eset \	view
Submission List			NAME		ODULE		REPORTING		ENTITY TYPE	ENTITY NAME		REFERENCE DATE		REMITTANCE DATE		RECEPTION DATE		ORIG./PARENT		PROCESSING	-	ACTIONS
File Vault		HLE	NAME		ODULE		CODE		ENTITY TYPE -	ENTITYNAME		REFERENCE DATE		REMITTANCE DATE		RECEPTION DATE		FILENAME		STATUS		ACTIONS
Validation Results				۲		۲		۲	T		۲	O DD / MM/ YYYY	۲	O DD / MM/ YYYY	۲		۲		۲		т	
Discussions	0	] TC.)	MYCOLL_TCY	с М	YMODULE		ENT1		Group	Entity 1		26/03/2020		30/04/2020		03/04/2020 11:02:5	i9	TC_MYCOLL_TCYCL	LE	File in Process		1
			IYCOLL_TCYC	L. M	IVMODULE		ENT1		Group	Entity 1		26/03/2020		30/04/2020		02/04/2020 15:16:4	15	C_MYCOLL_TCYCLE		File Rejected		1
		) C_N	YCOLL_REPO	Y. M	MODULE:	3	ENT1		Group	Entity 1		02/04/2020		10/05/2020		02/04/2020 14:50:4	15	C_MYCOLL_REPCY	ç	Completed		1
	E		YCOLL_REPO	Y M	MODULE	3	ENT1		Group	Entity 1		02/04/2020		10/05/2020		02/04/2020 14:47:2	21	C_MYCOLL_REPCY	c	Completed		1

Figure 104: File vault test submission

For zip file bulk uploads, make sure the file names all use the naming convention, i.e. they start with TC. The rest of the file upload process is the same.

## 3.2.17 Downloading data submitted

CASPER makes it possible to download all files that have been submitted. They can be downloaded from the submission list screen, but as explained in Section 3.2.11, this does not display rejected files or files submitted for optional reporting once the applicable reporting cycle has closed. To download files that are not visible, go to the file vault screen. Here you have access to all data submitted and you can download the files you require.

Please note that you may not be able to see files submitted in the past for a reporting cycle that is now closed. The collection designer can hide obsolete data in the submission list and the file vault.

Both screens offer two options for downloading files. To download a single file, simply select the file, click on "Actions" and then click on the "Download" option, as shown in Figure 105. The system downloads the file in its original submitted format.

										Submission Plat													9	
2				9	ubmissions											Us	er & I	loles						
0.00	File	Vault																						
lission List	Numb	per of Entries found: 18																	Down	iload Files		Export CSV		Reset V
ault ation Results		FILENAME	MODULE T		REPORTING CODE	ENTITY TYP		ENTITY NAME		REFERENCE DATE	-	REMITTANCE DATE		RECEPTION DATE		ORIG./PARENT FILENAME		PROCESSING		MODULE VERSION	τa	FILE SIZE (KB)	-8	ACTION
ssions		τ	upload	۲			۲		٣	0.007Mb/ YYYY	۲	0 00/ML/111	۲		۲		۲		۲	1	۲		۲	
5		C_DC2_UPLOAD_UPLOAD_ENTIT	UPLOAD		ENTITYI			ENTITY1		19/04/2021		31/05/2021		29/04/2021 10:19:39		100M8_25pererrors	xisx	Completed		1	-	100440		1
		C_DC2_UPLOAD_UPLOAD_ENTIT	UPLOAD		ENTITY1			ENTITY1		19/04/2021		31/05/2021		21/04/2021 15:24:14		100M8_25pererrors.	xisx	Completed		1		Download		
		C_DC2_UPLOAD_UPLOAD_ENTIT	UPLOAD		ENT/TY1			ENTITY1		19/04/2021		31/05/2021		21/04/2021 12:34:14		100MB_25pererrors.	xisx	Completed		1	ч	Cancel		(
		C_DC2_UPLOAD_UPLOAD_ENTIT	URIOND		ENTITY1			ENTITYS		19/04/2021		31/05/2021		21/04/2021 12:34:14		100MB_25pererrors.	-	Company				100440		_

Figure 105: Single file download

The second option lets you download one or more files. Simply select all the files you would like to download and then click on the "Download files" button highlighted in Figure 106. Please note that the system downloads all the selected files and puts them in a single zip file. To access the individual files, you will first need to unzip the file.

<b>Lodya</b>	File	e Va	ult																	
Submission List	Nu	mber of	Entries fo	und: 4												Do	wnload Files	Export	t CSV Reset View	N
File Vault Validation Results		FIL	ENAME	-	MODUL	E =	REPORTING	-	ENTITY TYPE	-	ENTITY NAME	-	REFERENCE DATE	-	REMITTANCE DATE	•	RECEPTION DATE 1	• =	ORIG./PARENT =	PROC
Discussions				т		٦		۲		۲		τ	OD/MV/YYYY	۲	O DD / MN / YYYY	۲		۲	τ	
Reports		C1	00005_F	RC1_M	MOD1		RE01		Company		test		13/04/2021		14/04/2021		11/05/2021 19:55:46	5	Test Submission.xlsx	Com
		C1	00005_R	RC1_M	MOD1		RE01		Company		test		13/04/2021		14/04/2021		04/05/2021 09:24:32	2	100005_MOD1_1_20	File F
	C	C1	00005_R	RC1_M	MOD1		RE01		Company		test		13/04/2021		14/04/2021		04/05/2021 08:31:23	7	Submission.xlsx	File F
		C_1	00005_R	C1_M	MOD1		RE01		Company		test		13/04/2021		14/04/2021		26/04/2021 19:36:54		SubmissionFile.xlsx	File F

Figure 106: Multiple file download

# **APPENDIX**

# Figures

Figure 1: CASPER landing page	4
Figure 2: Announcement preview sample	5
Figure 3: Announcement pop-up window	5
Figure 4: CASPER landing page	6
Figure 5: Receive notifications as emails	6
Figure 6: Manage email subscription screen	6
Figure 7: Email subscription configuration	7
Figure 8: Main navigation screen	8
Figure 9: Primary navigation pane	8
Figure 10: Secondary navigation pane	g
Figure 11: Access vertical navigation	g
Figure 12: Vertical navigation pane	g
Figure 13: Collapsible menu	10
Figure 14: Movable and resizable pop-ups	10
Figure 15: List components	11
Figure 16: Customisation panel	11
Figure 17: General options submenu	12
Figure 18: Adding toggle columns 1/3	13
Figure 19: Adding toggle columns 2/3	13
Figure 20: Adding toggle columns 3/3	13
Figure 21: Table headers	14
Figure 22: Filter button pop-up window	14
Figure 23: Filter pop-up types	14
Figure 24: Example of a text filter	15
Figure 25: Ascending sorting order	15

Figure 26: Descending sorting order	. 15
Figure 27: Multiple sorting	. 16
Figure 28: Multiple sorting with column in descending order	. 17
Figure 29: Example of an added row	. 17
Figure 30: Search panel details	. 17
Figure 31: Unchecked list entries	. 18
Figure 32: Checked list entries	. 18
Figure 33: Example of the "Actions" menu	. 19
Figure 34: Disabled Actions item	. 19
Figure 35: Export CSV functionality	. 19
Figure 36: ECB Identity Portal login screen	. 20
Figure 37: ECB Identity Portal landing page	. 21
Figure 38: Overview of the web user administrator journey	. 23
Figure 39: ECB Identity Portal login screen	. 24
Figure 40: User & Roles screen	. 24
Figure 41: Data collection assignment screen	. 25
Figure 42: Data collection unassignment screen	. 25
Figure 43: Data collection unassignment confirmation	. 25
Figure 44: Data collection assignment list	. 26
Figure 45: View user assigned entities	. 27
Figure 46: Data collection assignment list	. 27
Figure 47: Edit user assigned entities	. 27
Figure 48: User entity access report	. 28
Figure 49: Overview of the data submitter journey	. 29
Figure 50: ECB Identity Portal login screen	. 30
Figure 51: CASPER landing page	. 30
Figure 52: Notifications screen	. 31
Figure 53: Submission landing page	. 32
Figure 54: Submission list information	. 32

Figure 55: Submission List information pop-up	33
Figure 56: Submissions screen	34
Figure 57: Submission list screen showing reporting obligations	34
Figure 58: Parent-child relationship for resubmissions	38
Figure 59: Submission list for event-based data collections	38
Figure 60: Submissions screen for downloading a reporting template	42
Figure 61: Downloading a reporting template	42
Figure 62: Validation rules navigation	43
Figure 63: Validation rules screen	43
Figure 64: Validation rule pop-up window	44
Figure 65: Sample reporting template	45
Figure 66: Sample reporting template with completed data	45
Figure 67: Single file submission process	46
Figure 68: File Upload pop-up window	46
Figure 69: File upload confirmation screen	47
Figure 70: Event-driven submission	47
Figure 71: "Submit Event" pop-up	47
Figure 72: Zip file submission process	48
Figure 73: Zip file upload pop-up window	48
Figure 74: Encrypted and signed modules in submission list view	50
Figure 75: Attachment upload process	51
Figure 76: Attachment Upload pop-up window	51
Figure 77: Attachment upload confirmation screen	52
Figure 78: File processing: steps and final statuses	52
Figure 79: File vault screen	53
Figure 80: Navigating to validation results	56
Figure 81: Technical validation results screen	56
Figure 82: Business validation results screen	56
Figure 83: Validation results pop-up window	58

Figure 84: Navigating to the discussions screen	. 58
Figure 85: Discussions screen	. 58
Figure 86: Navigating to the discussion board screen	. 60
Figure 87: Reporting obligation panel and discussion topics panel	. 60
Figure 88: Comments panel	. 62
Figure 89: Add a comment to a discussion	. 62
Figure 90: Discussion topics panel – add attachment	. 63
Figure 91: Attachment upload screen	. 63
Figure 92: Discussions screen navigation	. 64
Figure 93: Discussions screen	. 65
Figure 94: Discussion board screen navigation	. 66
Figure 95: Reporting obligation panel and Discussion Topics panel	. 66
Figure 96: Comments panel	. 67
Figure 97: Add a comment to a discussion	. 68
Figure 98: Discussion topics panel – add attachment	. 68
Figure 99: Attachment upload screen	. 68
Figure 100: Reports screen	. 69
Figure 101: Report preview sample	. 69
Figure 102: Report pop-up screen	. 70
Figure 103: Testing submission pop-up	. 70
Figure 104: File vault test submission	. 70
Figure 105: Single file download	. 71
Figure 106: Multiple file download	. 71

## Tables

Table 1: Announcement pop-up fields	5
Table 2: Primary navigation items	8
Table 3: Secondary navigation items	9
Table 4: General options items	12
Table 5: Data submitter notifications	32
Table 6: Submission List default attributes	35
Table 7: Submission List additional attributes	36
Table 8: Obligation status types	37
Table 9: Submission list attributes for event-based data collections	39
Table 10: Optional attributes for submitted events	39
Table 11: Submission events list	40
Table 12: Submission events list additional attributes	41
Table 13: Validation rule attributes	44
Table 14: Available reporting obligation actions	46
Table 15: Naming convention attributes	49
Table 16: Upload limits	50
Table 17: File vault list default attributes	54
Table 18: File vault list – additional attributes	55
Table 19: File processing statuses	55
Table 20: Validation results screen attributes	57
Table 21: Validation results screen attributes	57
Table 22: Discussion screen default columns	59
Table 23: Discussion screen additional columns	60
Table 24: Reporting obligation panel attributes	61
Table 25: Discussion topics panel columns	61
Table 26: Comments panel default columns	62
Table 27: Comments panel: additional columns	62
Table 28: Discussion screen – default columns	65

Table 29: Discussion screen – additional columns	. 66
Table 30: Discussion topics panel columns	. 67
Table 31: Comments panel – default columns	. 67
Table 32: Comments panel: additional columns	. 67
Table 33: Abbreviations	. 75
Table 34: CASPER terminology	. 76

## Abbreviations and terminology

Abbreviation	Explanation
CSV	Comma separated value
XML	Extensible mark-up language
DAA	Delegated access administrator
ECB	European Central Bank
ESCB	European System of Central Banks
NCA	National competent authority
NCB	National central bank
RA	Reporting agent

### Table 33: Abbreviations

Term	Explanation
Collection owner	A collection owner may be any business area within the ECB. It sets the collection characteristics and receives the collected data.
Internal collection designer	Internal collection designers are ECB data collection experts who configure collections in the system. They configure the templates (called "modules" in CASPER), manage the reporting entities, configure business rules to validate the data collected and create reporting obligations for data submitters.
Comparison operator	A comparison operator compares the contents in a field either with the contents of another field or with a constant, and then returns a value based on whether the comparison is true and whether processing should proceed.

Term	Explanation
Data collection	A data collection brings together domain- specific information that a data consumer is interested in. For example, the Single Supervisory Mechanism may require data on non-performing loans to be provided by supervised institutions and specify the loan attributes to be reported. Data are mostly collected in cycles at an established frequency from a set of reporting institutions. Data collections are mostly structured in tables, for ease of comprehension by reporting institutions. They may change in extent and structure over time.
Data consumer	Data consumers are typically ECB/ESCB IT applications that provide business- specific functionalities and services for producing, disseminating and analysing data.
Delegated access administrator	Authenticated and authorised individuals who can assign application roles/access rights to ECB Identity Portal users within their organisation.
ECB Identity Portal	The access management system for web users.
Module	A set of tables to be submitted as a whole given the close relationships between the data points.
Received file	A file successfully uploaded to CASPER by an RA (either individually or as part of a zipfile). It may or may not result in a reporting obligation being fulfilled, depending on the technical validation.
Reception date	Date and time on which a file was received by CASPER.
Reference date	Date to which the reported data refer.

Term	Explanation
Remittance date	The deadline for receiving a mandatory submission.
Reporting agent	The user submitting the data.
Reporting cycle	The time window during which submissions of a specific set of modules for a reporting obligation are accepted by CASPER.
Reporting entity	The entity providing the reporting or the entity to which the reported data refer. This is usually a credit institution, but may also be an NCA/NCB, a combination of credit institutions and consolidation levels, an ad hoc entity (e.g. a bank's internal model, if a collection requires information at the level of individual models), or any other counterparty, i.e. experts affiliated to financial or non- financial institutions.
Reporting obligation	The obligation (or option) for a reporting entity to submit a file for a module for a given reference date. A reporting obligation is fulfilled if at least one submission completed meets the reporting obligation.
Reporting type	A reporting entity may submit data on behalf of a group as group head or for the company individually. If the same reporting entity code type and code are used for both reports, the type attribute enables the reporting levels to be distinguished.
Resubmission	If a submission contains validation errors, data submitters are expected to adapt their submission and resubmit. This is referred to as resubmission.

Term	Explanation
Submission	A submission is a file sent by a reporting entity for a given reference date. A submission is produced from a file received that has successfully passed the technical validation. A submission of structured data, once extracted, results in a list of observations with the associated validation results and metadata.
	A submission is always associated with a reporting obligation. More than one submission may be associated with the same reporting obligation (e.g. when there are resubmissions). Different submissions for the same reporting obligation are identified by ascending version numbers.
System administrator	The system administrators, also referred to as the CASPER Control Team, ensure compliance with governance processes, approve new data collections and launch the collection preparation process.
Validation	Validation is the process of checking files received based on the validation rules in the system. Validations are split into two categories: technical and business. Technical validation ensures a file received can be properly and reliably identified and the content extracted. Business validation can only be applied to data within a submission.
Validation results	Validation results show any errors identified in the validation process.

Table 34: CASPER terminology